Disability, Taxes, Transfers and the Economic Well-Being of Women

Bruce D. Meyer University of Chicago, NBER and AEI and Wallace K. C. Mok The Chinese University of Hong Kong

Abstract

We study the economic situation of disabled women and the role that taxes and transfers play in improving their economic circumstances, using 1968-2015 data from the Panel Study of Income Dynamics. We begin by documenting the trends in point-in-time disability rates of women as well as estimating the prevalence of disability over a woman's lifetime. We find that women are more likely than men to have experienced a disability through their mid-40s, but are less likely to have experienced a serious disability prior to retirement. The onset of disability for women is found to be associated with a fall in labor supply, family income and consumption. The fall varies with the degree of disability but tends to be smaller than that of disabled men, particularly for family income and consumption. Transfers, particularly Disability Insurance and SSI play a large role in cushioning the fall in income for disabled women. However, while the relative decline in income is smaller for women than men, a larger share of disabled women have very low absolute levels of income and consumption. Approximately half of the most disabled women receive SNAP benefits, pointing to the greater relative importance of means-tested benefits rather than social insurance for this group.

KEYWORDS: Disability, Income, Consumption, Poverty, Taxes, Transfers, Gender

NBER Tax Policy and the Economy Conference

September 14, 2017 Washington, D.C.

This research was supported by the U.S. Social Security Administration through grant #1 DRC12000002-02-00 to the National Bureau of Economic Research as part of the SSA Disability Research Consortium. The findings and conclusions expressed are solely those of the authors and do not represent the views of SSA, any agency of the Federal Government, or the NBER. We thank Pablo Celhay for research assistance and Nayoung Lee, Hans Van Kippersluis and Yin-Chi Wang for helpful conversations. We also thank Akshay Mangadu for excellent research assistance.

1. Introduction

There is a common view that rising disability insurance rolls are an alarming policy and budgetary issue (Autor and Duggan, 2006). However, research on the economic consequences of disability is less developed and has tended to focus on men. For example, Stephens (2001), Charles (2003), and Meyer and Mok (2017) examine only male household heads. This focus has occurred despite a rising share of the disabled who are women. According to the 2016 edition of the Annual Statistical Supplement to the Social Security Bulletin, the number of women receiving Social Security Disability Insurance (SSDI) in December of 2015 was about 4.3 million (excluding the children and spouses of disabled workers), a 43% increase relative to a decade earlier and a larger increase than for men (31%). In fact, there is a stronger case that disability rolls are inexplicably higher than in the past for women than for men (Liebman 2014). Recent data from the Survey of Income and Program Participation (SIPP) show that among the disabled under 65, 53% are women and this higher disability rate is observed across virtually all race groups (Brault, 2012).

The concern about rising disability rates also stems from the many studies that have investigated the moral hazard problems of disability insurance (Parsons, 1980; Autor and Duggan, 2002; von Wachter et al., 2011; Maestas et al., 2013). Yet a balanced assessment of the current disability insurance system requires an understanding of its benefits as well. Given that women have relatively lower earnings than men and the that SSDI benefits depend on past earnings, knowledge of whether the present system sufficiently guarantees a disabled woman's well-being is of vital importance to the design of the disability insurance system. Other work has shown that the nature and consequences of income loss differs between the genders (Weiss and Willis, 1997; Singleton, 2012). Maybe not surprisingly, disabled women are more likely to be living in officially defined poverty and rely more on means-tested public transfer than their male

1

counterparts.^{1,2} Furthermore, most of the existing evidence on women is based on crosssection evidence or short panels.

The purpose of this study is to fill this gap in the literature by providing multifaceted evidence on the association of disability with the well-being of women, and the role of taxes and transfers. We rely on 47 years of longitudinal data from the Panel Study of Income Dynamics (PSID) and other sources. We have several objectives: First, we study how point-in-time disability rates have evolved for women since 1980, as their labor market attachment has increased. Furthermore, we study the differences in lifetime disability prevalence between men and women. Second, following Meyer and Mok (2017), we examine the association between a woman's disability and her economic wellbeing over time as well as that of other household members. We measure well-being with a broad set of variables, including earnings, family income, food and housing consumption, and measures of poverty. We focus on the role played by taxes and transfers and compare how changes in most of these economic outcomes differ from those of disabled men.

Our present study differs significantly from the limited existing studies of disabled women. First, understanding that disability is often long term and persistent, though not always, a long panel data permits a better view of how disability affects individuals. Here, we employ the entire PSID panel data which covers a period of over 45 years. Second, understanding changes in the economic well-being of disabled women requires an examination of a large set of outcomes besides earnings and income. Our study looks at additional outcomes including consumption, and changes in wealth. Third, we account for the underlying differences between female heads and wives, given the former group of women is often economically deprived (Meyer and Sullivan 2012a, U.S. Census Bureau 2016).

¹ In 1992, the poverty rates for women with any work disability and severe work disability were 33.8% and 40.5%, respectively, and 24.2% and 31.2%, respectively, for men (U.S. Census Bureau, 1993). Poverty rates for non-disabled women and men were 12.1% and 8.1% respectively. Recent versions of this report do not provide estimates by gender and disability status.

² US Census Bureau (2011), Table 560, shows that in 2008, among women with work disabilities, 35.0%, 23.3%, and 62.0% received Social Security, Food Stamps and Medicaid, respectively. For men with work disability, these rates were 36.0%, 15.0%, and 68.9%, respectively.

This study has several key findings. Disability affects women very differently from men. Women are more likely to experience disability than men in their early working years, but the rates are similar at later working ages. A woman reaching the age of 56 has a 46.2% chance of ever experiencing a disability and a 11.26% chance that the experience will be of a serious form, characterized by the permanence and severity of the limitation. However, these rates are lower than those for men. While disabled women suffer a fall in key economic outcomes, including earnings, family income, and consumption following the onset of a disability, the fall depends on the nature of the disability and is on average smaller than for men. We find that by the tenth year after disability onset, an average disabled woman is estimated to suffer from a 25 percent drop in earnings, but only a 7.5 percent drop in after-tax post-transfer income and a 5.7 percent drop in food and housing consumption. Women suffering from a Chronic and Severe disability are estimated to experience an 82.5 percent drop in earnings, a 22.8 percent drop in after-tax post-transfer income, and a 13 percent drop in food and housing consumption. Our results on taxes and tax credits, and transfer program receipt indicate that these women with Chronic and Severe disabilities receive about \$7,700 in public transfers per year by the tenth year after disability onset, much of which comes from social security. Estimated federal income tax payments for the family fall substantially over the years of disability, reflecting the substantial reduction in earnings of the disabled women.

The rest of the paper is organized as follows: Section 2 describes the data and the methodology. Section 3 discusses the prevalence of disability from a lifetime perspective. Sections 4 examines changes in labor market outcomes for disabled women, while Section 5 examines income and income poverty outcomes. Section 6 examines consumption, consumption poverty and wealth. Section 7 provides further detail on the role of individual transfer programs and taxes, while Section 8 provides some robustness checks. Section 9 concludes the paper.

3

2. Data and Methodology

Our primary source of data is the PSID, a longitudinal dataset launched in 1968, with an initial sample of 4,802 US households and about 18,000 individuals. The survey has conducted interviews annually since 1968 and bi-annually since 1997. Split-offs, such as divorcees or children forming their own family, are followed and interviewed. Besides demographic information, the survey provides comprehensive data on transfer program receipt, earnings, income, food, and housing consumption. The longitudinal nature of these data allow an investigator to track economic outcomes for an individual over a long period of time. As of the 2015 wave, data from 77,223 individuals had been collected.

In this study, we use the entire PSID panel, covering 1968–2015. However, the PSID survey does not collect the same information from every individual and the questionnaire changes from time to time. In particular, the survey initially focused on the family head, normally defined as the principal male family member, and only later treated female spouses in a parallel fashion. Because of this restriction, we must focus on female household heads and wives, but we find that these two groups constitute the vast majority of adult women. How such a data structure affects our sampling frame is discussed below.

Defining Disability and Disability Rates

The key question we use to determine an individual's disability status is: "*Do you have any physical or nervous condition that limits the type or amount of work you can do?*" While the use of such a self-reported response is controversial in disability studies, we have no good alternative in the PSID.³ After determining the presence of a work-limiting condition, the interviewer asks a severity question to determine the extent to

³ See the discussion in Meyer and Mok (2014). Past work that argue self-reported disability/health indicators are endogenous includes Baker et al. (2004), Kreider (1999), Kreider and Pepper (2007). Studies that argue self-reported indicators are close to exogenous include Stern (1989), Dwyer and Mitchell (1999), Benítez-Silva et al. (2004). Campolieti (2002) finds that using self-reported disability carries a "downward bias" relative to a case of instrumenting disability with health measures in labor supply equations.

which this condition limits the individual's work capability.⁴ We follow the same strategy as Meyer and Mok (2014) and group the responses to the severity question into two categories: "*Severely Disabled*" (for those who respond "*can do nothing*," "*completely*," "*a lot*," or "*severely*" to these severity questions) and "*Not Severely Disabled*" (for those who respond "*just a little*," "*somewhat*," "*not limiting*," or "*not at all*" to these severity questions).

Such disability questions were asked of the family head fairly consistently in the survey. However, wives were asked these questions only in 1976 and in the 1981–2015 waves of the survey. Given such a data structure, we focus on two groups of women: female household heads and married women. These two groups of women constitute about 80% of all women ages 22-61 in the United States, and usually a slightly higher percentage of all disabled women in the same age range. Hence, we view these two groups as providing a good approximation to the patterns for all women in the United States.⁵

Table 1 shows the weighted point in time disability rates for current female heads and wives ages 18–61 during the PSID survey period 1968–2015. These rates are high and are a few percentage points higher than those found for male household heads (Meyer and Mok, 2017). Although this female disability rate rose from 12.8% in 1981 to 14.2% in 2015, the disability rate fluctuated quite a bit over this period.⁶ We also observe that female heads are more likely to suffer from work limitations than wives, with an average difference of about 5 percent points in the 1980s, 4 percent points in the 1990s and about 7 percentage points since 2000. Regarding the fraction of disabled women reporting a severe work limitation, one should note that the severity question changed in 1986 to ask about limits to "work you can do" instead of about any limits (see the online appendices of Meyer and Mok, 2017); so the revised question was more

⁴ Meyer and Mok (2014) have shown that the response to such a question has a high correlation with selfreports of various physical and health limitations.

⁵ Using the Current Population Survey (CPS) Annual Demographic File/Annual Social and Economic Supplement (ASEC), we find that female household heads and wives of householders comprise about 82% of women aged 18–61 years during 1989–2012. The rest are women who live with their parents or siblings (who are the householder) or other related and unrelated householders (such as friends).

 $^{^{6}}$ There is some evidence that the rate has systematically increased over time, as a regression of the rate on a constant and year gives a time coefficient of 0.00046 (s.e. 0.00020, p-value 0.0034). We obtain similar estimates when we use the Prais-Winsten AR(1) procedure.

restrictive in defining a severe disability. Not surprisingly, this change had the effect of decreasing the share of disabled women recorded as severely disabled between 1985 and 1986. However, after 1988 there was a noticeable upward trend in the share of women reporting a severe work limitation and this is more apparent for female household heads than for wives.

One might be concerned that men and women differ in how they characterize a given work limitation, which would cast doubt on the possibility of comparing genders as is done in some subsequent sections of this study. There are two points worth noting. First, questions about wives are usually answered by their husbands in the PSID, unless they are incapacitated, although for female-headed households, these questions are answered by the female heads themselves. Second, the 1986 and 1999–2015 waves of the PSID also asked questions about limitations in several physical activities (such as walking, bathing) and doctor-diagnosed health problems. We have examined the gender differences in these indicators for those with no current disability, those whose disability is not severe, and those whose disability is severe. In general, we find no large differences between the genders in the association between disability and these physical limitations or diagnosed health problems, and thus consider the interpretation of these disability questions to be similar for the two genders.

Sample Construction

The principal strength of the PSID is its longitudinal nature, but because of its long time frame and small sample size, using a balanced panel would be too restrictive. We use an unbalanced panel with a different number of years for each woman, but we impose restrictions similar to those of Meyer and Mok (2017) to ensure sufficient information for each individual in our major analyses. First, we require that the individual be a head or wife for at least six surveys, four of which must be consecutive, while the individual is age 22–61. Second, we delete those individuals with missing key demographic information (race, marital status, age, or education).⁷ Third, since the focus of the paper is the changes in economic circumstances of women following disability

⁷ To the extent possible, we impute the missing values of key demographic variables using the nearest available wave of data.

onset, knowledge of when an individual became disabled is essential. Determining the year of limitation onset for the disabled sample requires combining information from multiple years of data. A valuable feature of the PSID available only for heads (but not wives) in the 1969–1975 and 1978 waves is a retrospective question asking when a work limitation began. For those female heads disabled on or before 1978, we use the responses to this question to determine their year of disability onset.⁸ We require that individuals who first reported having a disability after 1978 report no limitations in the two consecutive survey years immediately prior to the year in which they first reported having a work limitation.⁹ We should emphasize that year of onset is not precise; the vast majority of disabilities are conditions that evolve slowly over time rather than accidents where there is a clear before and after period.

Our focus is on disabilities that begin during the working years; accordingly, we exclude those whose onset age is under 18 or above 56.¹⁰ To obtain sufficient information after onset, we require that a disabled individual in our sample take part in the survey for a minimum of three years during the ten years after disability onset. This restriction is important to determine disability persistence and severity groups (introduced below). Due to the restrictions that we impose in selecting our sample, we slightly understate the extent of work limitations, as discussed further below. The application of these restrictions results in a primary sample of 7,423 women, 2,175 (29.3%) of whom are classified as ever having been disabled.

Classifying Disability

Meyer and Mok (2017) point out that treating the disabled as a single group can be misleading; disaggregation based on the permanence and severity of a disability

⁸ Some individuals may have more than one response due to the panel nature of the data. Since the responses to these questions were coded in intervals (except in the 1978 survey, when the exact number of years is given), we determine the intersection of the intervals given by these questions and take the earliest year within the intersection as the year of disability onset, similar to Meyer and Mok (2014).

⁹ For example, if an individual first reports having a limitation in 1983, then the year of onset would be 1983 if the individual had no limitations in 1981 and 1982. Since there is only one interview per year, we also choose the year of onset to be the year including the midpoint in time of adjacent interviews.

¹⁰ Our main estimation sample includes the person-year observations prior to disability onset for those who became first disabled after age 56 as they form part of the implicit comparison group for the disabled.

reveals substantial heterogeneity in outcomes. We also follow this approach here. The *One-Time Disabled* are those who report a disability once but do not report a disability again during the next ten years. The *temporarily disabled* are those who have one or two positive limitation reports during the ten years after disability onset, while the *Chronically Disabled* are those who have three or more positive limitation reports during the ten years.

Since the severity questions are asked nearly every year, we rely on average severity in the paper. Specifically, we define the *severity ratio* as the fraction of time the individual reports she is *Severely Disabled* in the year of onset and the subsequent ten years after onset.¹¹

We combine the two disability dimensions in our main analyses by splitting the *Chronically Disabled* into two groups. Hence, this classification yields four groups of interest – *One-time, Temporary, Chronic-Not Severe* (with a severity ratio under 0.5) and *Chronic-Severe* (with a severity ratio over 0.5) which we collectively call the *Extent of Disability groups*.¹² In their first disability spell, 652 (30%) of the sample of disabled women are classified as *One-Time*, 572 (26%) as *Temporary*, 629 (29%) as *Chronic-Not Severe*, and 322 (15%) as *Chronic-Severe*. Much of our analysis focuses on the *Chronic-Severe* group, because this group fares unusually poorly after disability. Using the SIPP and Detailed Earnings Records of the Social Security Administration, Singleton (2014) has also shown that the disabled with self-reported *Work-Preventing* limitations do not experience a rebound in their earnings, even if their SSDI application is rejected.¹³ This result suggests that their disability is real and serious.¹⁴

Table 2 provides descriptive statistics for our sample. On average, the disabled are about four years older than the non-disabled, are less likely to be white, and are less

¹¹ Individuals who never responded to the severity question in this 11-year period (year of onset and the subsequent 10 years) are dropped from the main analyses.

¹² In the case where exactly half of the responses indicate severe disability (a severity ratio of 0.5), we classify the disabled head based on the first severity report.

¹³ Although our *Chronic-Severe* group and the *Work Preventing* group in Singleton's study are not identical, we expect there to be a large overlap between these two groups.

¹⁴ Many authors argue that using a self-reported definition of disability in estimating the effect of health on labor supply would suffer from justification bias (which would overstate the estimated effect) and attenuation bias (which would bias the effect towards zero) but that these two biases may cancel each other out (Bound, 1991). The evidence of Singleton (2014) seems to suggest that the justification bias is small for the worst-off self-reported group.

likely to be married but to participate in more waves of the survey. Among the disabled, the *Chronic–Severe* group is generally older on average, much less likely to be white, and much less likely to be married. Nevertheless, the four disabled groups have on average participated in a similar number of interviews, though the two chronic groups have responded more often since the year of disability onset.

Methodology

Following past studies of outcomes with longitudinal data, we follow the popular event-study approach and estimate individual fixed effect regressions of two forms:

(1)
$$y_{it} = \alpha_i + \gamma_t + X_{it}\beta + \sum_g \sum_k \delta_k^g A_{kit}^g + \varepsilon_{it},$$

(2)
$$y_{it} = \exp(\alpha'_i + \gamma'_t + X_{it}\beta' + \sum_g \sum_k \delta'_k A^g_{kit}) + \varepsilon'_{it}$$

Model (1) is a standard linear regression with individual fixed effects, while model (2) is a Poisson model with individual fixed effects. Which model we use for each outcome is explained below. The variable y_{it} is an outcome of interest for person or family *i* in year *t* (such as family food and housing consumption); α_i (α_i ') is an individual fixed effect and γ_t (γ_t ') is an indicator variable for year *t*; while X_{it} is a set of time-varying explanatory variables, including marital status, state of residence, age and age squared, education, and the number of children. We also include interactions of these variables for some dependent variables (see the data appendix for more details). To account for the underlying life-cycle differences between the two groups of women, we include age and age-squared interacted with an indicator variable for a woman who is classified as a female head.¹⁵ The variable A_{kit}^g is an indicator that is equal to one if, in year *t*, a disabled individual *i* belonging to disability group *g* and is *k* years from the year of disability onset, and ε_{it} (ε_{it}) is a potentially serially correlated error term. Our coefficients of interest are δ_k^g (δ'_k^g), measuring the change in the dependent variable during the *k*-th year

¹⁵ We classify a non-disabled woman as a female head if we observe her to be a female head for at least half of the time in the PSID survey during which she was age 22-61. For the disabled, we look at the fraction of years in the five years from onset when the woman was head. The disabled woman is classified as head if the fraction is at least 0.5.

from disability onset for a disabled individual belonging to disability group *g*, relative to her outcome in the baseline period, the time more than five years prior to disability onset. The inclusion of individual fixed effects removes all time-constant unobservables of the person or family. Because of the inclusion of control variables such as age, our estimates account for how a woman's outcomes would have evolved over time. We estimate (1) whenever it is more natural to examine how disability affects the *level* of a dependent variable, while we estimate (2) when we want to study how disability affects the *percentage change* in the dependent variable.¹⁶ Although it is popular to estimate a loglinear version of model (1) when studying the percentage changes in an outcome, a Poisson model is better if the dependent variable has a value of zero for many observations (which makes it difficult to take logarithms), as is the case for annual earnings, since many women do not work.

In our analysis, all monetary values are reported in 2016 dollars, adjusting for inflation using the Consumer Price Index Research Series Using Current Methods (CPI-U-RS).

3. Lifetime Disability Prevalence of Women

While most research reports the percentage of women who are disabled (based on a point-in-time self-report or program-based definition), a more important statistic for insurance purposes (Baily 1977; Chetty 2006) is arguably the lifetime prevalence of disability (i.e., the probability of *having had* a prolonged disability any time prior to a given age). The PSID, with its longitudinal structure and long history, is suited to calculate this rate.

Ideally we would like to capture this measure over a person's entire work lifetime, but even with 47 years of data we cannot do it for most sample members. With individuals aging in and out of their working years, it is important to use only individuals who have had a long enough time period in the sample so that the experience we are recording is approximately their lifetime probability of ever having had the various types of disability, some of which are long term. To do so, we focus on the sample of female

¹⁶ The percentage change is obtained by exponentiating the estimated coefficient and subtracting one.

household heads and wives present in at least 1984–1994 who responded to the disability questions at least 10 times by this period.¹⁷ We choose 1994 as the last year in this period of potential onset so we have sufficient information to classify the disability of a woman whose disability started that year. We also account for the potential worsening of a condition and thus a change in a person's disability group classification. We classify a person according to the worst disability they experience.

Table 3 shows these lifetime disability prevalence rates. In theory, these rates should be monotonic with age, but they are not here because of sampling variation and the unbalanced nature of the PSID panel. By the time a woman reaches age 30, she is estimated to have a 25.6% chance of ever being disabled, though the chance of ever suffering from a *Chronic–Severe* disability is less than 1%. As she ages, the lifetime disability prevalence increases. By the time she reaches 40 years of age, there is a 31% chance of her ever having suffered from a disability, with a 4% chance of a *Chronic–Severe* disability. These rates increase rapidly as a woman enters her 50s. By the time she reaches 56 years of age, there is a 46% chance that she will have suffered from a disability during her working years, with a 11% chance of ever experiencing a *Chronic–Severe* disability. As shown below, the *Chronic–Severe* group experiences much worse outcomes than the average disabled. Coupled with the fact that there is a 20% chance of such a disability for a women by age 60, one should not think of membership in this group as that unusual.

We compare these lifetime disability prevalence rates with those of men shown in Meyer and Mok (2017). For ease of comparison, Figure 1 shows the lifetime disability prevalence rates (any and *Chronic–Severe*) for men and women. We see that the probability of having had a disability (of any type) is generally higher for women than for men before reaching the mid-forties. The pattern is slightly reversed afterwards: by age 60, the probability of ever having had a disability is very similar for men and women. Regarding the prevalence of a *Chronic–Severe* disability, the rates are fairly similar for men and women prior to reaching age 48 but, from then on, we observe a relatively rapid

¹⁷ Specifically, we select these women in the 1984–1994 period from the person–year data format. For a person–year observation in this subsample, we further require the individual to have 10 or more years of disability information by this year.

increase in the chance of a man ever experiencing a *Chronic-Severe* disability. By age 60, a male household head is estimated to have a 26% probability of ever having experienced a *Chronic–Severe* disability, while it is 20% for our female sample. These numbers provide alternative evidence that the higher point-in-time disability rates observed for women using cross-sectional data may not reveal the entire picture.

4. Disability and the Hours of Work, Employment, and Earnings of Women

To begin our study of the well-being of working-age disabled women, it is natural to start with the relationship between disability and labor force involvement. We first examine how disability is associated with annual hours worked. Estimating (1) using annual hours of work as the dependent variable and treating the disabled as a single group yields the results shown in column 1 of Table 4. In the year of disability onset, it is estimated that annual hours of work drop by about 169 on average (relative to the period more than five years prior to onset) and this fall continues through ten years after onset, when the drop is estimated to be about 300 hours. In terms of labor supply at the extensive margin, we examine the raw fraction of women not working in a given year (defined as working zero hours in the year). Column 2 of Table 4 shows that on average there is about a 10 percentage point increase in the fraction of women not working over the 11-year period from onset to ten years after.

Turning our attention to results that disaggregate the disabled, we report in columns 1 and 2 of Table 5 the change in hours of work and the raw fraction not working for the most disabled group, the *Chronic-Severe* group.¹⁸ We plot these changes for all disability groups along with the average disabled in Figures 2 and 3 for ease of comparison. For the *Chronic-Severe* group, annual hours of work are estimated to drop sharply by 457 hours by the year of disability onset. By the tenth year after onset, the estimated drop has more than doubled, reaching an estimated 1,166 hours with more than three-quarters of such disabled not working. One can see the sharp differences in the figures, where the drop in the annual hours of work for the *Chronic-Severe* group by the tenth year after onset is almost quadruple that of the average disabled and triple that of

¹⁸ Detailed results for the other disability groups are available from the authors upon request.

the less disabled *Chronic-Not Severe* group. For the *One-Time* and *Temporary* groups, the drops are relatively small, as one would expect, given their relatively mild disabilities.

To examine annual earnings, we estimate model (2) given the prevalence of zero earnings. Column 3 of Table 4 shows the estimated changes for the average disabled while the corresponding results for the *Chronic-Severe* group are reported in column 3 of Table 5. These results are illustrated in Figure 4. The average disabled woman is estimated to suffer a 13.3 percent decline in annual earnings by the year of onset, and the drop accelerates to 23.3 percent by the fifth year after disability onset, and stays around this level for the next 5 years. An examination of the results suggests that most of this drop is attributed to the *Chronic-Severe* group, with an estimated drop of about 42.8 percent by the year of onset and a massive drop of 83 percent by the tenth year after onset. One should not be surprised by this result, since more than three-quarters of the disabled in the *Chronic-Severe* group are not working by this time. One should also note that treating the disabled as a single group, as done in many past studies, could be misleading. This issue is especially important for future research on the disabled using survey data.

We also note a modest drop in earnings for the *Chronic-Severe* group before the year of onset. This fall prior to self-reported onset is observed in other research employing an event-study framework (Singleton, 2012; Meyer and Mok, 2017). A plausible explanation is that an individual may take a while to declare she is disabled after a health shock.¹⁹ As noted in Section 2, it is best to think of disability as a process not an event, despite our emphasis on year of onset. It may therefore be preferable to focus on the years around onset rather than its exact point in time when interpreting these results.

Comparing these results with those of male household heads (Meyer and Mok, 2017), we find the decline in annual work hours for disabled women as a whole is somewhat smaller. The drop for disabled women is on average about 70 percent of that of disabled men. The increase in non-work after disability onset is quite similar between

¹⁹ For example, the individual may first experience a modest level of pain that distracts her at work, but not enough for her to declare that she is work limited. The yearly format of the survey (biannual format in recent years) can also result in an approximation to the year of disability onset.

the two genders. From onset to the tenth year after onset, it is 10 percentage points for women and 12 percentage points for men. The earnings decline is also very similar, at 13–15 percent by the year of onset and 25 percent by the tenth year after onset. However, some notable differences emerge as we switch our focus to the various disability groups, especially the Chronic-Severe. Disabled men in this category suffer from a much larger drop in annual hours of work (about 300 hours more) than their female counterparts, which is mostly attributed to the greater increase in non-work for these men. The fraction of these disabled men working zero hours roughly quadruples, from 16 percent by the year of onset to 66 percent by the tenth year after onset. For the *Chronic-Severe* disabled women though, this fraction only doubles, from about 35 percent to 78 percent. In terms of the earnings change, both men and women in this disability category experience similar falls, specifically, 39–43 percent by the year of onset and 77–83 percent by the tenth year after onset.

5. Income before Taxes/Transfers, and Income Poverty

While the drop in earnings for the disabled is large, whether it translates into a large decline in material well-being for the individual requires an examination of other non-labor sources of income, especially public benefits and social insurance. Other private mechanisms, such as spousal earnings (to be discussed below) and private insurance benefits, may also be important in mitigating the decline in material well-being associated with the drop in earnings due to disability. We estimate specification (2) again with several measures of family income, thus illustrating the important role of social insurance and government benefits in mitigating falls in income. We report results for 1) Pre-tax and pre-transfer family income, 2) After-tax income prior to transfers, 3) After-tax income with non-SSA transfers (i.e. exclude SSI and OASDI), 4) After-tax income with only SSA transfers, and 5) After-tax income with all transfers. These measures progressively show how different arrangements help cushion the drop in earnings. The

later measures includes many public transfer benefits, and account for the underreporting of these benefits, though in a rough way.²⁰ Taxes are estimated using TAXSIM.²¹

The results for the disabled as a whole are reported in columns 4-8 in Table 4, while the results for the *Chronic-Severe* group are shown in the corresponding columns in Table 5. For ease of comparison, the results for all of the groups and the average disabled are plotted in Figure 5 (After-tax Pre-Transfer Income), Figure 6 (After-tax Post-Transfer Income. Results for the Pre-tax pre-Transfer Income, After-tax income with non-SSA transfers and After-tax income with SSA transfers are displayed respectively in the Appendix Figures A1-A3. For the average disabled, the drop in income is small by the year of onset, but the estimates quickly become more negative over the course of disability. As earnings continue to drop for the average disabled, pretax, pre-transfer income is dropping too. As expected, such drops are smaller in magnitude due to family income pooling and non-labor income. In most cases, the drop in such income is half of the drop in earnings alone. Including taxes and tax credits, reduces the fall by a few percentage points. Comparing the results for the next two aftertax income measures, one can also see the increasing importance of targeted programs of SSI and SSDI in the sense that these SSA administered programs play a greater role in reducing the income fall over the course of disability, compared with the non-SSA programs such as UI and Food Stamps/SNAP.

In sum, after-tax income prior to transfers is estimated to drop by 9 percent by the year after onset for all disabled, while the drop is less than 5 percent when all public transfers are included. By the tenth year following onset, after-tax income without public transfers is estimated to have fallen by 10.9 percent, and after-tax income with public transfers by 7.5 percent. Relative to the drop in earnings, private arrangements diminish the drop by 56 percent and, coupled with public transfers and taxes (changes presented in Figure 6), more than 70% of the drop in earnings is cushioned.

²⁰ These public transfer benefits include AFDC/TANF, Food Stamps/SNAP, Veterans Benefits, Social Security, Supplemental Security Income, and the estimated value of subsidized housing. To account for underreporting, we scale up the amounts of these benefits using the reporting rates shown in Meyer, Mok and Sullivan (2015).

²¹ The current calculations only include federal income taxes. We will add payroll taxes and state income taxes in the next version.

Across the disability groups, we again observe substantial differences and the drop in income for the average disabled is again mostly due to the Chronic-Severe group. By the year of onset, the Chronic-Severe are estimated to experience a drop of 19.6 percent in their family after-tax income before accounting for public transfers. The inclusion of public transfers reduces the drop to 15.3 percent. The role of private and public transfers becomes increasingly significant over time. By the tenth year after onset, after-tax income prior to transfers is estimated to drop by about 39.8 percent, but accounting for public transfers reduces this to 22.8 percent. Figure 7 shows the changes in public transfer receipts over the course of disability, obtained by estimating (1) with the amount of public transfer received as the dependent variable (the estimates are also reported in Table 7 (average disabled) and Table 8 (for the Chronic-Severe). Indeed, for *Chronic-Severe* women, the value of public transfers received is estimated to quintuple over the course of their disability. Starting from about \$1,200 annually in the years prior to onset, transfers increase to about \$5,800 annually by two years after onset, and further increase to about \$7,700-9,200 per year by 6-10 years after disability onset as can be seen in Table 8.

Contrasting these results with those of disabled male household heads, the family income drop for disabled women is generally slightly smaller than for men. Without accounting for transfers, families with a disabled woman suffer from a drop in income over the course of her disability that is 3–6 percentage points smaller than families with a disabled male head and such a difference persists even after we account for the various types of public benefits received. The differences are more noticeable when we focus on *Chronic-Severe* disabled men and women, especially in the later stage of the disability spell. Without public transfers, the drops in after-tax income for families with disabled women are much smaller than for their male counterparts. By the year of onset, the drop for families with a disabled female head is about 20 percent, while it is about 23 percent for families with a disabled man. By the fifth year after onset, the drop is about 39 percent for women, and 49 percent for men. By the tenth year after onset, the drop is about 40 percent for public transfers: By the tenth year after onset, families with women are slightly smaller when we account for public transfers: By the tenth year after onset, families with

16

such disabled women suffer from a 20 percent decline in income, compared to 28 percent for families with a disabled man.

While the standard economic insurance framework suggests a focus on changes in outcomes, the level of deprivation is an important indicator in itself. We calculate the poverty rate, measured as the fraction of families with income below the thresholds prescribed by the U.S. Census Bureau, which differ by family size and structure. Figure 8 shows the poverty rate of families in different disabled groups over the years of disability. To arguably better capture the degree of material deprivation, we use after-tax income with public transfers as the basis for determining poverty status.²² The *Chronic-Severe* group is much more likely to be living below the poverty line, even in the years prior to disability onset, but the poverty rate for this group still increases to more than 30 percent during the initial years after disability onset. By the tenth year after disability onset, the poverty rate for the *Chronic-Severe* group is still about 28 percent. In comparison, the rate for the average disabled woman is only about 15 percent.

While other outcomes following disability for women tend to be better than for men, the poverty rate of women is substantially higher. The poverty rate of Chronic-Severe disabled women is nearly ten percentage points higher than that of men in the years after onset, averaging over thirty percent, compared to just over twenty percent for men. These results indicate that, on an absolute scale, disabled women fare poorly even though their household income does not fall proportionately as much as it does for men.

6. Consumption, Poverty and Wealth

While income is the most readily available measure of material well-being, a substantial literature suggests that consumption does a better job of capturing a household's living standard. From a theoretical perspective, material well-being is more

²² Note that the standard income measure used by the U.S. Census Bureau is pre-tax and does not include in-kind transfers, such as food stamps, or the possible underreporting of government benefits. While there are good conceptual reasons to account for taxes and transfers, given errors in reporting and imputation we should caution that it is not clear that such adjustments will lead to a measure that is more highly correlated with deprivation (Meyer and Sullivan, 2012).

directly tied to current consumption than to current income, since income is subject to transitory fluctuations caused by events such as job or family composition changes. Living standards may remain unaffected despite large income changes through saving and borrowing (Cutler and Katz, 1991; Poterba, 1991). From a practical perspective, measuring disposable income by accounting for taxes can be complicated in survey data. In addition, consumption may be more accurately reported than income for those who are disadvantaged, possibly due to the many small irregular sources of income received by this group (Edin and Lein, 1997; Meyer and Sullivan, 2003, 2011). Furthermore, consumption is more closely associated with other measures of well-being for the disadvantaged (Meyer and Sullivan 2003, 2011).

The PSID has several variables that can be used to provide a measure of consumption, including expenditures on food eaten at home and outside the home, rent, and home value. In this section we define food consumption as the sum of expenditures on food eaten in and outside the home and the face value of food stamps. We also construct a housing consumption variable as the sum of rent paid, six percent of the current home value (if the family is an owner), and the rental equivalent for those in subsidized housing.²³ We define family consumption as the sum of food and housing consumption.

Food Consumption

Estimating (2) with food expenditure as the dependent variable yields the results shown in columns 9 of Table 4 (for the average disabled woman) and Table 5 (for the *Chronic-Severe* group). These results are also displayed in Figure 9. Although there is some indication that food consumption on average drops for women suffering from a disability, this drop is always under 3 percent and usually much smaller throughout the 10 years following disability onset. For the *Chronic-Severe* disabled (Table 5) the drop is estimated to be larger, averaging around 3 percent after onset, but it is not significantly different from zero in any single year. It tends to rise over time though, and by the tenth

²³ See Meyer and Mok (2017) for details on the housing expenditure variable.

year after onset, the drop in food expenditure is estimated to be 6.9 percent (with a standard error of 4.5).

These rather small drops in food consumption with disability observed above may be due to a shift from food eaten outside the home to food eaten at home as income falls (and preferences potentially change). Although not reported, we have also investigated the changes in these two food measures. For food eaten at home, we observe that the changes are mostly small and are all statistically insignificant, even for the Chronic-Severe group. Turning to expenditures on food eaten outside the home, the drop for the average disabled is somewhat larger compared with those for food eaten at home, but still averages only about 4.5 percent, and is in most years statistically insignificant. For the Chronic-Severe disabled however, the drop in food away is substantial. By the year of onset, food away from home is estimated to drop by about 20.2 percent (statistically significant at the 1 percent level) and the decline continues, reaching about 30 percent by 2-4 years after onset (all significant at the 1% level). Some recovery is observed thereafter (though the estimates are noisy), but by the tenth year after onset the drop remains high, at 30.3 percent (significant at the 1% level). Based on these results, there seems to be a large degree of substitution toward eating at home following a severe disability to a female household head or wife.

Food plus Housing Consumption

The most comprehensive measure of consumption that we can construct in the PSID is the sum of food and housing, which accounts for almost half of expenditures for the average family. Figure 10 shows the results for this broader consumption measure (individual coefficients reported in column 9 of Table 4 for the average disabled and in Table 5 for the *Chronic-Severe*). The drop for the average disabled is about 3.8 percent by the year of onset, increasing to about 3-6 percent in the 1-3 years after disability onset and then increasing further to about 6-8 percent 5-10 years after disability onset. Although these estimates are nearly all statistically significant, they are still fairly small in size, especially when compared with those for the *Chronic-Severe* group. The Chronic-Severe disabled experience a 8.4 percent drop in food plus housing consumption

19

by the year of onset, and then a further fall to about 10.4 percent below the pre-disability standard by the third year after onset. Consumption stays at this low level through the 10^{th} year after onset, with most of the individual year estimates significantly different from zero.

Compared with those of men, the drop in food plus housing consumption and its components for our female sample is generally smaller among the average disabled and the *Chronic–Severe*. For the average disabled woman, the drop in food consumption is typically less than a third that experienced by men. For food plus housing consumption the drop is about two-thirds of that of men. For the *Chronic-Severe* disabled the differences between women and men are even more evident. For food consumption, the fall for men is about four times larger. In the case of the broader food plus housing measure the difference is about a factor of two, with a fall for women of about 11 percent by 6-10 years after onset compared to 25 percent for men. Based on these comparisons, the *Chronic-Severe* disabled group is faring very badly, although the families of such disabled women are doing relatively better than those of similarly disabled men. However, this finding may not be that surprising. Women's earnings, in the period of data we use, usually constitute a smaller fraction of family income than the earnings of men. Therefore, the loss in earnings due to a woman's disability has a smaller impact on family income and therefore family consumption on average than a man's disability.

Consumption Poverty

Besides the income poverty measure, we have also calculated a consumption based poverty rate for disabled women. We define a family as living below consumption poverty if the family's food plus housing consumption is less than half of its designated Census income poverty threshold since food plus housing is typically about half of overall consumption. Figure 11 shows the consumption poverty rate for the various disabled groups over the years of disability. Relative to the income poverty rates reported earlier, these consumption poverty rates are typically a few percentage points higher. For the Temporary and the Chronic Not-Severe groups the consumption poverty rates are slightly higher upon disability onset, but return to their pre-disability level quite

20

quickly in the years after. The consumption poverty rate for the Chronic-Severe group is higher than the other groups even before disability onset, a result consistent with that of income-poverty. The trends in the two poverty measures are also quite similar, as we see a rise starting from the second year prior to onset until around the seventh year when the rate begins to fall. In the years after onset, both poverty measures average about 30 percent for the Chronic-Severe group, a very high rate.

Changes in Net Wealth

To complete the picture of the economic circumstances of the different disabled groups and to examine how the pieces fit together, we report the degree of dissaving. Wealth data come from the 1984, 1989, 1994, and 1999–2015 waves of the PSID. We linearly interpolate family wealth between years from the available data. We define net wealth as total wealth, including home equity. The last two rows of Table 6 show median net wealth in the year of disability onset and in the 6-10 years after. Median Net Wealth at onset is smallest for the Chronic-Severe group, and continues to be so in the long run after disability onset. We do not observe any evidence of dissaving at the family level over the period of a woman's disability, with a rise in wealth similar to that of the other disabled and nondisabled groups. We should emphasize though that we have many fewer observations on wealth than we do for other outcomes and the data are likely to be less accurately recorded.

7. The Role of Individual Transfer Programs and Taxes

Our array of results in Section 4 shows that while disabled women suffer from large drops in earnings, their drops in income and especially in consumption are relatively modest, suggesting the important role of public transfer receipts. Table 6 shows the receipt rates of various transfers for the different disabled groups in the 6-10 years after disability. About 43 percent of families with such Chronically and Severely disabled women receive benefits from the Social Security Administration (OASI, SSDI or SSI), with 35 percent of them receiving social security in the form of OASI and SSDI. It is also surprising to see that about 43 percent of such disabled women receive food stamps (SNAP). This suggests that many women may not have earned enough during their pre-disabled years and are therefore ineligible for SSDI or entitled to only very small benefits. It should be pointed out that our Chronic-Severe group consists of women who are more likely to be household heads, black, older, and high-school dropouts. Relative to male disabled household heads in the same category (Meyer and Mok, 2017), Chronic-Severe disabled women are less likely to receive social security (14 percentage point lower in likelihood), but are more likely to receive SSI and especially Food Stamps/SNAP (almost 20 percentage points higher in likelihood).

To see the relative changes in the amount of various transfers received by a disabled woman, we have also estimated equation (1) with each type of benefit as the dependent variable in turn. The results are shown in Table 7 (all disabled) and Table 8 (Chronic-Severe). These results are displayed in Appendix Figures A4-A9. We do not observe large changes in benefits received from programs such as UI and AFDC/TANF that have eligibility requirements unrelated to disability. For workers' compensation (WC) (Figure A7), we observe an increase of about \$1,000 in the later years of disability for the Chronic-Severe, although the estimates are very noisy. The picture is very different for the two benefits that target the permanently disabled. For SSI, there is a very noticeable increase from the year of onset for the two Chronic groups. The increase for the Chronic-Severe group is particularly dramatic, reaching about \$2,500 (in 2016 dollars) by the tenth year after disability onset. For Social Security, the increase for the Chronic-Severe group is equally noticeable, reaching about \$5,500 by the tenth year after disability. Relative to the Chronic-Severe disabled male heads, these Chronic-Severe disabled women receive more in SSI and less in Social Security in the long run. Coupled with the receipt rates reported in Table 6, it may be that the relatively lower earnings of women implies SSI payments are higher and social security benefits are lower, as the amount of SSI benefit awarded is independent of an applicant's prior earnings but goes down with higher social security benefits. In addition, eligibility for SSDI requires the applicant to have "worked long enough and recent enough" and have earned sufficient social security credits (that are based on prior annual earnings). The relatively lower

earnings of women may render them ineligible for SSDI, and more likely to be SSI eligible in the event of disability.

Consistent with the results on earnings, estimated federal income tax liabilities are lower, especially so for the Chronic-Severe group given their large decline in earnings.

8. Additional Results

Split by SSA benefit receipt

The results above indicate the importance of several public programs, in particular social security. We might be interested in seeing how some of the key economic outcomes differ between those who receive SSA administered benefits and those who do not. Given the stringent SSA disability requirements, it is useful to examine how disabled women not covered by these programs fare after disability onset.

We estimate the regressions for the key economic outcomes of annual hours of work, after-tax post-transfer income and food plus housing consumption. We split the *Chronic-Severe* group into those who receive SSDI or SSI benefits (at the family level) more than half of the time over the ten years after disability onset (*SSA recipients*) and those who do not (*SSA non-recipients*), and combine the other extent of disability groups together. The results are illustrated in Appendix Figures A11 (Annual hours of work), A12 (After-tax post-transfer income) and A13 (Food plus Housing Consumption). For these two Chronic-Severe groups, the SSA recipients have larger relative drops in hours of work than the non-recipients. This result is expected as a current recipient cannot engage in substantial gainful activity (for 2017, this is defined as having monthly earnings above \$1,950). Recipients also experience slightly larger drops in income in the short run following disability onset. In terms of food plus housing consumption, the drops are similar between these two groups. One should not attach a causal relationship here; it is entirely possible that these differences are due to differences between those who decides to apply for SSI or SSDI and those who do not.

Cohort Differences

Given the evolution of the SSDI and SSI programs and the increasing labor market participation of women over time, we examine whether the material circumstances of the disabled have changed over the period of data we have. To do so, we split the disabled into two samples: those who were first disabled before 1990 (the early group), and those disabled later (the recent group). We estimate the outcome regressions on these two samples separately. The results are reported in a series of Appendix Tables (Tables A5-A8 for those who were first disabled before 1990, and Tables A9-A12 for those who were disabled after 1990). For most outcomes the changes are similar but we shall highlight a few importance differences. We find that for the average disabled, those disabled in the recent period have larger drops in hours of work (usually above 100 hours per year) as well as larger percentage drops in earnings. At the extensive margin of labor supply, disabled women in the two periods are similar in the short run, but diverge beginning with the fourth year after onset, with the recent disabled women showing greater labor force attachment than the earlier group. In terms of benefit receipts, the later disabled women have greater receipt of Social Security benefits, and food stamps/SNAP especially in the 7-10 years after disability onset.

Turning to the more important Chronic-Severe group, the drop in their economic outcomes is generally larger for those who became disabled after 1990. For earnings, the recent Chronic-Severe disabled experienced drops that are about 20 percentage points higher than their earlier counterparts. The drop in After-tax post-transfer income is also about 15 percentage point larger for the recent disabled group. For consumption, again the drop for the recent disabled group is typically several percentage points higher over the later years of the window. Relative to the case of all disabled women, the Chronic-Severe group in the recent period seem to have experienced greater hardship during the path of disability than those who were disabled earlier. In terms of benefit receipt, the recent Chronic-Severe disabled group receives greater such transfers in the first few years after disability onset relative to their early counterparts, but the difference narrows starting with the eighth year after onset. Social Security income plays a larger role for the recent Chronic-Severe disabled than other types of transfers. All these results are

24

consistent with the rising female labor force participation and earnings over the last two decades leading to a greater loss of income upon disability, but higher receipt of SSDI that doesn't quite compensate for the greater earnings loss.

Heads v. Wives

We examined the differences in outcomes for heads of households compared to wives. We found that heads experienced greater pre-tax and pre-transfer declines in income, but received larger transfers that tended to eliminate any noticeable differences in other outcomes between the two groups.

9. Conclusions

The problem of a growing disabled population and its associated public expenditures has attracted the attention of academics and policy makers. Most of the current debate, however, has focused on the moral hazard problem of disability insurance and less so on its economic benefits. In addition, the few studies on the well-being of the disabled have mostly focused on men. This study provides a panoramic view of how disability affects the well-being of women. While disability is associated with reduced earnings and labor supply for women similar to its relationship for men, its association with reduced family income and family consumption is somewhat smaller than for disabled men. However, the poverty rate for disabled women is much higher than for men, particularly for the most severely disabled.

In models such as Chetty (2006), the decline in consumption with disability is one of the key factors determining the optimal generosity of disability benefits. In this context, the much smaller decline in consumption for women than men is a factor that suggests that there is less of a case for more generous disability insurance for women than for men. On the other hand, the much high rate of poverty for disabled women than for disabled men is a factor that suggests a greater case for programs that target the disabled poor such as Supplemental Security Income. Of course, there are other

25

important factors in setting policies such as the differences in moral hazard under alternative policies for the disabled.

References:

- Autor, David H. and Mark G. Duggan. 2006. "The Growth in the Social Security Disability Rolls: A Fiscal Crisis Unfolding." *Journal of Economic Perspectives*, 20(3): 71-96.
- Autor, David H. and Mark G. Duggan. 2003. "The Rise in the Disability Rolls and the Decline in Unemployment." *Quarterly Journal of Economics*, 118(1): 157-205.
- **Baily, Martin Neil.** 1977. "Unemployment Insurance as Insurance for Workers," *Industrial and Labor Relations Review*, 30, 495-504.
- **Baldwin, Robert and Sharon Chu.** 2006. "A Death and Disability Life Table for Insured Workers Born in 1985." *Actuarial Note*. No. 2005.6. Baltimore MD: Office of the Chief Actuary, Social Security Administration.
- Benítez-Silva, Hugo; Moshe Buchinsky; Hiu Man Chan; Sofia Sheidvasser and John Rust. 2004. "How Large Is the Bias in Self-Reported Disability?" *Journal of Applied Econometrics*, 19(6): 649-70.
- Ben-Shalom, Y., Moffitt, R.A., and Scholz, J. K. 2012. An Assessment of the Effectiveness of Anti-Poverty Programs in the United States. In P.N. Jefferson (Ed.), The Oxford Handbook of the Economics of Poverty. Oxford: Oxford University Press.
- Bound, John, Burkhauser, Richard.V., and Austin Nichols. 2003. "Tracking the household income Of SSDI and SSI applicants," *Research in Labor Economics* 22:113-15.
- Bound, John. 1991. "Self-Reported Versus Objective Measures of Health in Retirement Models." *Journal of Human Resources*, 26(1):106-138
- Brault, Matthew W. 2012. "Americans with Disabilities: 2010." *Current Population Reports*. No. P70-131. Washington, DC: US Department of Commerce and Bureau of Census.
- Campolieti, Michele. 2002. "Disability and the Labor Force Participation of Older Men in Canada." *Labour Economics*, 9:405-32
- Charles, Kerwin Kofi. 2003. "The Longitudinal Structure of Earnings Losses among Work-Limited Disabled Workers." *Journal of Human Resources*, 38(3): 618-46.
- Chetty, Raj. 2006. "A General Formula for the Optimal Level of Social Insurance." *Journal of Public Economics*, 90(10-11): 1879-901.
- Cutler, David M. and Lawrence F. Katz. 1992. "Rising Inequality? Changes in the Distribution of Income and Consumption in the 1980s." American Economic Review Papers and Proceedings, 82(2): 546-51
- **Duggan, Mark, Melissa Kearney and Stephanie Rennane.** 2015. "The Supplemental Security Income Program." In *Economics of Means-Tested Transfer Programs in the United States, Volume II.* Edited by Robert A. Moffitt, NBER, University of Chicago Press.
- Dwyer, Debra Sabatini and Olivia S. Mitchell. 1999. "Health Problems as Determinants of Retirement: Are Self-rates Measures Endogenous?" *Journal of Health Economics*, 18:173-93
- Edin, Kathryn, and Laura Lein. 1997. Making Ends Meet: How Single Mothers Survive Welfare and Low-Wage Work. New York: Russell Sage Foundation.

- Kreider, Brent. 1999. "Latent Work Disability and Reporting Bias." *Journal of Human Resources*, 34(4): 734-69.
- Kreider, Brent and John V. Pepper. 2007. "Disability and Employment: Reevaluating the Evidence in Light of Reporting Errors." *Journal of the American Statistical Association*, 102: 432-41.
- Liebman, Jeffrey. 2014. "Understanding the Increase in Disability Insurance Benefit Receipt in the United States." *Journal of Economic Perspectives* 29, pp. 123-150.
- Low, Hamish and Luigi Pistaferri. 2015. "Disability Insurance and the Dynamics of the Incentive Insurance Trade-Off." *American Economic Review* 105, pp. 2986-3029.
- Maestas, Nicole, Kathleen Mullen, and Alexander Strand. 2013. "Does Disability Insurance Receipt Discourage Work? Using Examiner Assignment to Estimate Causal Effects of SSDI Receipt," *American Economic Review*, 103(5):1797-829.
- Meyer, Bruce D. and Wallace K. C. Mok. 2013. "The Economic Consequences of Disability in America: Evidence from the PSID." in Unexpected Life Cycle Events, edited by Ken Couch, Mary Daly, and Julie Zissimopoulos. Stanford University Press: Palo Alto, CA.
- Meyer, Bruce D. and Wallace K. C. Mok. 2017. "Disability, Earnings, Income and Consumption." NBER Working Paper No. 18869, March 2013, Revised June 2017.
- Meyer, Bruce D.; Wallace K. C. Mok and James X. Sullivan. 2009. "The Under-Reporting of Transfers in Household Surveys: Its Nature and Consequences." NBER Working Paper No. 15181, July 2009.
- Meyer, Bruce D. and James X. Sullivan. 2003. "Measuring the Well-Being of the Poor Using Income and Consumption." *The Journal of Human Resources*, 38(Special Issue on Income Volatility and Implications for Food Assistance Programs): 1180-220.
- Meyer, Bruce D. and James X. Sullivan. 2011. "Viewpoint: Further Results on Measuring the Well-Being of the Poor using Income and Consumption." *Canadian Journal of Economics* 44 (1) February 2011, 52-87.
- Meyer, Bruce D. and James X. Sullivan. 2012a. "Winning the War: Poverty from the Great Society to the Great Recession". *Brookings Papers on Economic Activity*, Fall 2012, 133-200.
- Meyer, Bruce D. and James X. Sullivan. 2012b. "Identifying the Disadvantaged: Official Poverty, Consumption Poverty, and the New Supplemental Poverty Measure," *Journal of Economic Perspectives* Summer 2012, 111-136.
- Moffitt, Robert A. 2015. "The Deserving Poor, the Family, and the U.S. Welfare System." *Demography* 52: 729-749.
- Mok, Wallace K.C.; Bruce D. Meyer; Kerwin Kofi Charles and Alexandra C. Achen. 2008. "A Note on "the Longitudinal Structure of Earnings Losses among Work-Limited Disabled Workers." *Journal of Human Resources*, 43(3): 721-8.
- **Parsons, Donald O.** 1980. "The Decline in Male Labor Force Participation." *Journal of Political Economy*, 88(1): 117-34.
- Poterba, James M. 1991. "Is the Gasoline Tax Regressive?" in *Tax Policy and the Economy 5*, David Bradford, ed. Cambridge, MA: MIT Press, 145-164.

- Scholz, J.K., Moffitt, R. and Cowan, B. 2009. Trends in income support. In: M. Cancian and S. Danziger, (Eds.), Changing poverty, changing policies, Washington, D.C.: Russell Sage Foundation.
- Singleton, Perry. 2014. "The Dynamic Relationship between Disability Onset, Earnings, and Disability Application and Receipt" Working Paper.
- Stephens, Melvin, Jr. 2001. "The Long-Run Consumption Effects of Earnings Shocks." *Review of Economics and Statistics*, 83(1): 28-36.
- Stern, Steven. 1989. "Measuring the Effect of Disability on Labor Force Participation." Journal of Human Resources, 24(3): 361-95.
- **von Wachter, Till, Jae Song, and Joyce Manchester.** 2011. "Trends in Employment and Earnings of Allowed and Rejected Applicants to the Social Security Disability Insurance Program." *American Economic Review*, 101(7): 3308-29.
- Weiss, Yoram and Robert J. Willis. 1997. "Match Quality, New Information, and Marital Dissolution." *Journal of Labor Economics*, 15(1):S293-329
- **U.S. Bureau of the Census**. 1993. Poverty in the United States, 1992. Current Population Reports, Series P60-185. Washington, DC, U.S. Government Printing Office.
- **U.S. Bureau of the Census**. 2011. Statistical Abstract of the United States: 2012 (131st edition). US Department of Commerce, Economics and Statistics Administration.
- **U.S. Bureau of the Census. 2016**. Income and Poverty in the United States, 2015. Current Population Reports, Series P60-256. Washington, DC, U.S. Government Printing Office.

	All Fen	nale Heads a	and Wives	Female Heads			Wives		
Year	N	%	% of	N	%	% of	N	%	% of
		Disabled	Disabled		Disabled	Disabled		Disabled	Disabled
			that are Severe			that are Severe			that are Severe
1968			Severe	1.063	0.208	0.459			Severe
1969				983	0.230				
1970				987	0.227				
1971				1,007	0.260				
1972				1,041	0.211	0.519			
1973				1,060	0.189	0.512			
1974				1,100	0.178	0.497			
1975				1,142	0.182	0.454			
1976	4,004	0.129	0.353	1,151	0.192	0.461	2,853	0.110	0.290
1977				1,175	0.212	0.559			
1978				1,190	0.196	0.557			
1979				1,243	0.206	0.506			
1980				1,267	0.190	0.533			
1981	4,578	0.128	0.487	1,326	0.187	0.526	3,252	0.106	0.461
1982	4,699	0.120	0.515	1,397	0.191	0.540	3,302	0.092	0.494
1983	4,624	0.129	0.417	1,404	0.175	0.427	3,220	0.110	0.411
1984	4,685	0.140	0.355	1,422	0.170	0.415	3,263	0.127	0.320
1985	4,751	0.151	0.361	1,452	0.175	0.437	3,299	0.142	0.325
1986	4,743	0.113	0.274	1,453	0.118	0.339	3,290	0.111	0.242
1987	4,798	0.140	0.257	1,491	0.165	0.233	3,307	0.128	0.272
1988	4,824	0.161	0.238	1,495	0.168	0.311	3,329	0.158	0.202
1989	4,820	0.149	0.276	1,472	0.175	0.300	3,348	0.137	0.260
1990	6,246	0.151	0.282	1,915	0.177	0.327	4,331	0.140	0.255

 Table 1: Disability Rates of Women, 1968-2015

Notes: The sample includes female heads and wives ages 22-61. The rates are weighted using family weights.

	All Fen	nale Heads a	and Wives	Female Heads			Wives		
Year	N	% Disabled	% of Disabled	Ν	% Disabled	% of Disabled	N	% Disabled	% of Disabled
			that are Severe			that are Severe			that are Severe
1991	6,126	0.153	0.263	1,863	0.182	0.287	4,263	0.139	0.247
1992	6,421	0.144	0.248	1,980	0.170	0.269	4,441	0.130	0.234
1993	6,520	0.134	0.247	2,122	0.162	0.284	4,398	0.121	0.225
1994	6,980	0.134	0.277	2,290	0.168	0.306	4,690	0.122	0.262
1995	6,703	0.137	0.301	2,205	0.168	0.358	4,498	0.122	0.262
1996	5,526	0.148	0.289	1,863	0.177	0.327	3,663	0.133	0.263
1997	4,528	0.140	0.311	1,433	0.159	0.335	3,095	0.130	0.297
1999	4,700	0.144	0.296	1,524	0.171	0.338	3,176	0.133	0.272
2001	4,957	0.146	0.261	1,594	0.188	0.317	3,363	0.127	0.222
2003	5,222	0.143	0.345	1,763	0.214	0.382	3,459	0.109	0.310
2005	5,330	0.140	0.322	1,820	0.191	0.370	3,510	0.115	0.284
2007	5,450	0.146	0.358	1,923	0.186	0.441	3,527	0.126	0.297
2009	5,615	0.152	0.341	2,077	0.190	0.421	3,538	0.131	0.279
2011	5,615	0.144	0.334	2,200	0.174	0.384	3,415	0.126	0.294
2013	5,588	0.160	0.358	2,236	0.200	0.400	3,352	0.136	0.321
2015	5,473	0.142	0.426	2,290	0.173	0.523	3,183	0.123	0.341

Table 1: Disability Rates of Women, 1968-2015 (continued)

Notes: The sample includes female heads and wives ages 22-61. The rates are weighted using family weights.

				Extent of D	Disability	
	Non- Disabled	All Disabled	One- Time	Temporary	Chronic Not Severe	Chronic Severe
Age at Disability Onset		37.1	34.9	36.7	37.7	41.2
Age	35.7 (7.5)	(10.2) 39.4 (7.9)	(9.7) 36.5 (5.9)	(10.2) 37.8 (7.0)	(10.3) 40.8 (8.0)	(9.5) 45.2 (8.9)
White	0.630 (0.483)	0.585 (0.493)	0.655 (0.476)	0.615 (0.487)	0.588 (0.493)	0.385 (0.487)
Married	0.697 (0.384)	0.577 (0.414)	0.696 (0.375)	0.634 (0.394)	0.517 (0.421)	0.349 (0.401)
Number of Years in Survey (ages 22-61)	14.0 (8.2)	20.2 (8.3)	19.9 (8.3)	20.0 (8.5)	21.7 (8.0)	18.1 (8.1)
Highest Level of Education - High School	0.310 (0.462)	0.317 (0.466)	0.310 (0.462)	0.332 (0.471)	0.312 (0.464)	0.289 (0.454)
Highest Level of Education – College	0.559 (0.497)	0.434 (0.496)	0.540 (0.499)	0.472 (0.500)	0.383 (0.487)	0.252 (0.435)
Years in Survey after Onset		12.8 (7.8)	10.6 (6.0)	11.0 (7.2)	15.2 (8.1)	15.7 (9.2)
Number of Consecutive Positive Limitation Reports		1.404 (3.865)		0.313 (0.554)	2.334 (4.410)	4.366 (6.777)
Number of Valid Reports of Disability Status from Onset to the 10th Year after Onset		6.948 (2.480)	6.606 (2.483)	6.470 (2.527)	7.671 (2.258)	7.078 (2.478)
Number of Positive Limitation Reports from Onset to the 10th Year after Onset		2.654 (2.709)		1.407 (0.492)	4.979 (1.986)	5.699 (2.138)
Severity Ratio (up to the 10th Year after Onset)		0.265 (0.361)	0.155 (0.362)	0.221 (0.316)	0.141 (0.165)	0.810 (0.174)
Age in the Last Interview	46.4 (12.3)	55.7 (13.5)	52.0 (12.1)	53.6 (13.2)	58.6 (13.6)	61.3 (13.3)
Number of Observations	5,248	2,175	652	572	629	322

Table 2: Sample Means for Non-disabled and Disabled Women

Notes: Standard Deviations are in parentheses. See text for the construction of the sample.

				Extent of Disability				
		Any	Currently	One-		Chronic-Not	Chronic-	
Age	N	Disability	Disabled	Time	Temporary	Severe	Severe	
30	476	0.2551	0.0402	0.0941	0.0933	0.0596	0.0082	
		(0.0267)	(0.0113)	(0.0188)	(0.0180)	(0.0136)	(0.0049)	
32	733	0.2660	0.0949	0.0783	0.0904	0.0815	0.0159	
		(0.0212)	(0.0136)	(0.0124)	(0.0146)	(0.0127)	(0.0054)	
34	867	0.3034	0.0865	0.0997	0.0949	0.0740	0.0347	
		(0.0202)	(0.0119)	(0.0128)	(0.0137)	(0.0111)	(0.0076)	
36	935	0.2902	0.1101	0.1048	0.0610	0.0866	0.0378	
		(0.0191)	(0.0133)	(0.0130)	(0.0101)	(0.0120)	(0.0076)	
38	930	0.3276	0.0986	0.0956	0.0765	0.1081	0.0475	
		(0.0195)	(0.0123)	(0.0120)	(0.0108)	(0.0132)	(0.0093)	
40	825	0.3066	0.1011	0.1050	0.0615	0.0986	0.0416	
		(0.0203)	(0.0131)	(0.0134)	(0.0101)	(0.0134)	(0.0089)	
42	731	0.3377	0.1009	0.1017	0.0860	0.1132	0.0368	
		(0.0224)	(0.0136)	(0.0142)	(0.0132)	(0.0158)	(0.0083)	
44	606	0.3344	0.1211	0.0944	0.0680	0.1326	0.0394	
		(0.0245)	(0.0171)	(0.0153)	(0.0133)	(0.0182)	(0.0094)	
46	503	0.3327	0.1256	0.0946	0.0639	0.1164	0.0578	
		(0.0262)	(0.0183)	(0.0170)	(0.0142)	(0.0170)	(0.0130)	
48	400	0.3218	0.1402	0.0607	0.0771	0.1261	0.0579	
		(0.0287)	(0.0215)	(0.0154)	(0.0158)	(0.0203)	(0.0139)	
50	394	0.3220	0.1353	0.0524	0.0936	0.1254	0.0506	
		(0.0295)	(0.0216)	(0.0140)	(0.0181)	(0.0210)	(0.0128)	
52	359	0.4190	0.1978	0.0810	0.1249	0.1304	0.0826	
		(0.0332)	(0.0270)	(0.0186)	(0.0228)	(0.0225)	(0.0174)	
54	345	0.4139	0.1780	0.0628	0.1257	0.1625	0.0630	
		(0.0346)	(0.0258)	(0.0179)	(0.0246)	(0.0261)	(0.0136)	
56	337	0.4625	0.2000	0.0749	0.1088	0.1669	0.1120	
		(0.0362)	(0.0279)	(0.0197)	(0.0232)	(0.0265)	(0.0209)	
58	291	0.6176	0.2816	0.1247	0.1537	0.1858	0.1534	
		(0.0378)	(0.0339)	(0.0263)	(0.0286)	(0.0295)	(0.0255)	
60	273	0.6472	0.3070	0.1203	0.1084	0.2229	0.1956	
		(0.0382)	(0.0358)	(0.0266)	(0.0244)	(0.0325)	(0.0305)	

Table 3: Lifetime Prevalence of Disability among Women

Notes: This table reports for each age the fraction of the sample members who have had a disability by the specified age, the fraction of individuals who are currently disabled, and the fraction for whom a given disability type is their most severe disability to date. For this table we only use data from 1984-1994 and individuals with at least 10 years of disability data prior to the specified age. The fractions are weighted using family weights. Standard errors are in parentheses. See text for details.

Table 4: Changes in	Economic Outcome	es relative to baseline.	All-Disabled Women
Table I. Changes in		s i chative to baseline,	I III DISabica women

Year from Onset	Hours	Fraction not working	Earnings	Pre-tax Pre- transfer Income	After-tax Income without Transfers	After-tax income with non-SSA Transfers	After-tax income with SSA Transfers	After Tax Income with all Transfers	Food Consumption	Housing Consumption	Food plus Housing Consumption
-5	23	0.23	1.38	-2.1	-1.7	-1.27	-1.8	-1.13	0.87	-2.81	-1.28
	(20)	(0.01)	(2.12)	(1.85)	(1.58)	(1)	(1.52)	(1.46)	(1.18)	(1.83)	(1.16)
-4	9	0.23	-0.59	-1.02	-0.98	-0.41	-0.35	0.11	0.43	-4.19**	-2.27*
	(22)	(0.01)	(2.31)	(1.91)	(1.62)	(2)	(1.67)	(1.60)	(1.07)	(1.59)	(1.09)
-3	-27	0.22	-2.34	-3.3	-2.67	-2.36	-2.17	-1.98	0.46	-2.79	-1.55
	(22)	(0.01)	(2.27)	(1.98)	(1.74)	(2)	(1.80)	(1.74)	(1.32)	(2.19)	(1.43)
-2	-21	0.24	-1.24	-1.63	-2	-1.83	-1.62	-1.12	-1.54	-3.62	-3.15*
	(24)	(0.01)	(2.46)	(2.20)	(1.83)	(2)	(1.91)	(1.81)	(1.15)	(2.10)	(1.38)
-1	-105**	0.25	-6.76**	-4.65*	-5.03**	-3.88*	-4.35*	-3.07	-0.77	-3.32	-2.58
	(24)	(0.01)	(2.44)	(2.26)	(1.78)	(2)	(1.84)	(1.76)	(1.21)	(2.06)	(1.38)
0	-169**	0.26	-13.30**	-4.12	-4.85	-3.83	-4.23	-2.97	-0.63	-5.63**	-3.75**
	(27)	(0.01)	(2.54)	(3.47)	(2.63)	(3)	(2.82)	(2.68)	(1.43)	(2.00)	(1.43)
1	-286**	0.33	-21.20**	-9.28**	-8.95**	-6.58**	-7.72**	-4.97**	-0.32	-4.93*	-3.37*
	(26)	(0.01)	(2.44)	(2.32)	(1.88)	(2)	(1.96)	(1.89)	(1.59)	(2.22)	(1.53)
2	-243**	0.33	-19.55**	-10.75**	-10.45**	-7.22**	-8.33**	-5.55**	-1.11	-6.61**	-4.90**
	(28)	(0.01)	(2.57)	(2.28)	(1.85)	(2)	(1.94)	(1.90)	(1.22)	(2.09)	(1.44)
3	-255**	0.34	-21.88**	-11.63**	-10.76**	-8.44**	-7.91**	-6.01**	-1.67	-7.85**	-5.75**
	(28)	(0.01)	(2.48)	(2.05)	(1.76)	(2)	(1.81)	(1.74)	(1.32)	(2.21)	(1.52)
4	-244**	0.34	-21.01**	-8.76**	-8.64**	-7.99**	-6.66**	-5.68*	-2.71*	-7.53**	-6.32**
	(30)	(0.01)	(2.61)	(2.83)	(2.26)	(2)	(2.34)	(2.23)	(1.31)	(2.05)	(1.45)
5	-290**	0.35	-23.31**	-12.16**	-11.51**	-10.37**	-8.76**	-7.51**	-2.33	-6.97**	-5.94**
	(29)	(0.01)	(2.65)	(2.57)	(2.10)	(2)	(2.28)	(2.17)	(1.46)	(2.28)	(1.59)
6	-265**	0.35	-22.72**	-9.36**	-9.96**	-7.83**	-6.63**	-5.62*	-2.76	-8.99**	-7.24**
	(31)	(0.01)	(2.75)	(2.79)	(2.18)	(2)	(2.31)	(2.19)	(1.42)	(2.26)	(1.64)
7	-303**	0.36	-21.17**	-6.71	-7.51**	-6.69**	-4.13	-4.05	-1.18	-9.66**	-7.28**
	(30)	(0.01)	(2.82)	(3.48)	(2.63)	(3)	(2.66)	(2.51)	(2.04)	(2.74)	(1.90)
8	-291**	0.35	-22.66**	-10.61**	-10.10**	-9.51**	-7.99**	-7.22**	-1.63	-9.55**	-7.50**
	(33)	(0.01)	(2.89)	(2.58)	(2.13)	(2)	(2.11)	(2.01)	(1.44)	(2.42)	(1.73)
9	-311**	0.37	-22.77**	-9.40**	-8.59**	-8.27**	-5.99**	-5.58*	-2.79	-8.96**	-7.43**
	(33)	(0.01)	(3.19)	(2.71)	(2.34)	(2)	(2.30)	(2.21)	(1.63)	(2.52)	(1.78)
10	-302**	0.36	-24.96**	-11.26**	-10.90**	-10.31**	-8.07**	-7.48**	-1.67	-6.85*	-5.71*
	(36)	(0.02)	(3.13)	(3.59)	(2.81)	(3)	(2.78)	(2.86)	(1.72)	(3.26)	(2.27)

		Fraction		Pre-tax Pre-	After-tax	After-tax income	After-tax	After Tax	Food	Hausing	Food plus
Year from		not		transfer	Income without	with non-SSA	income with	Income with	Congumation	Consumption	Housing
Onset	Hours	working	Earnings	Income	Transfers	Transfers	SSA Transfers	all Transfers	Consumption	Consumption	Consumption
-5	-90	0.26	-7.52	-8.12	-7.74	-6	-5.06	-5.48	-0.65	-1.08	-1.83
	(58)	(0.03)	(5.68)	(4.63)	(4.32)	(4)	(4.35)	(3.71)	(4.24)	(4.43)	(3.08)
-4	-78	0.27	-10.23	-10.01*	-8.45*	-8.84*	-4.48	-5.48	-5.29	-7.59*	-7.49**
	(64)	(0.04)	(6.06)	(3.89)	(3.65)	(3)	(3.54)	(3.13)	(3.28)	(3.76)	(2.69)
-3	-222**	0.24	-16.57**	-10.01*	-7.16*	-8.62*	-5.34	-6.62*	4.50	-8.11*	-2.61
	(65)	(0.03)	(6.24)	(3.98)	(3.57)	(3)	(3.53)	(3.20)	(6.23)	(3.84)	(3.85)
-2	-186*	0.27	-18.04**	-5.76	-4.45	-9.64*	-0.55	-5.64	1.20	-9.36*	-5.70
	(72)	(0.03)	(6.64)	(4.49)	(4.16)	(4)	(3.99)	(3.50)	(4.34)	(4.34)	(3.23)
-1	-385**	0.32	-31.57**	-20.99**	-19.26**	-18.06**	-16.34**	-14.99**	2.42	-5.21	-3.41
	(70)	(0.03)	(6.27)	(4.12)	(3.73)	(4)	(3.71)	(3.38)	(4.08)	(4.54)	(3.10)
0	-457**	0.35	-42.80**	-21.22**	-19.61**	-18.70**	-16.53**	-15.30**	-2.26	-10.56*	-8.41**
	(75)	(0.03)	(5.45)	(4.29)	(3.95)	(4)	(3.94)	(3.57)	(3.93)	(4.54)	(3.07)
1	-901**	0.59	-69.53**	-32.13**	-29.81**	-23.77**	-22.14**	-17.05**	0.21	-9.65*	-6.97*
	(71)	(0.03)	(4.21)	(3.71)	(3.50)	(3)	(3.51)	(3.24)	(4.32)	(4.46)	(3.19)
2	-934**	0.60	-70.14**	-36.50**	-34.63**	-25.66**	-24.07**	-18.14**	-2.94	-11.47*	-9.34**
	(80)	(0.03)	(4.46)	(4.32)	(4.10)	(4)	(4.11)	(3.62)	(4.09)	(4.61)	(3.17)
3	-998**	0.68	-76.83**	-38.97**	-36.71**	-29.26**	-24.88**	-18.95**	-7.40	-9.92	-10.36**
	(72)	(0.03)	(3.42)	(3.86)	(3.71)	(4)	(3.72)	(3.30)	(4.05)	(5.54)	(3.45)
4	-1,017**	0.68	-74.64**	-33.26**	-31.50**	-26.82**	-18.01**	-15.31**	-5.72	-14.78**	-12.79**
	(76)	(0.03)	(3.87)	(4.40)	(4.20)	(4)	(4.21)	(3.50)	(4.03)	(4.77)	(3.14)
5	-1,082**	0.70	-79.68**	-40.76**	-38.74**	-33.88**	-23.93**	-21.02**	-6.17	-9.96	-11.15**
	(71)	(0.03)	(3.35)	(3.95)	(3.75)	(4)	(3.72)	(3.40)	(3.93)	(5.66)	(3.60)
6	-1,092**	0.72	-79.63**	-33.96**	-32.51**	-24.15**	-17.78**	-11.59*	-0.92	-14.74**	-10.42**
	(78)	(0.03)	(3.81)	(5.78)	(5.43)	(5)	(5.69)	(4.72)	(4.21)	(4.80)	(3.43)
7	-1,118**	0.75	-76.21**	-39.97**	-38.87**	-32.45**	-21.22**	-18.24**	-7.07	-6.46	-9.52
	(72)	(0.03)	(5.42)	(4.55)	(4.06)	(4)	(4.16)	(3.68)	(4.40)	(12.10)	(6.39)
8	-1,108**	0.71	-78.01**	-40.38**	-37.62**	-30.25**	-21.76**	-17.38**	-3.90	-13.71**	-10.88**
	(80)	(0.03)	(4.00)	(4.68)	(4.51)	(4)	(4.67)	(4.29)	(3.88)	(5.26)	(3.29)
9	-1,168**	0.79	-83.92**	-43.48**	-41.39**	-37.38**	-23.63**	-23.19**	-2.06	-14.50**	-11.23**
	(75)	(0.03)	(3.14)	(4.95)	(4.52)	(4)	(4.43)	(3.74)	(5.51)	(5.07)	(3.87)
10	-1,166**	0.78	-82.51**	-43.13**	-39.82**	-38.08**	-20.37**	-22.81**	-6.88	-15.03**	-13.09**
	(80)	(0.03)	(3.42)	(4.26)	(4.16)	(4)	(4.60)	(3.66)	(4.49)	(5.20)	(3.70)

Table 5: Changes in Economic Outcomes relative to baseline, Chronic-Severe Disabled Women

Table 6

			Extent o	f Disability	
	All Disabled	One-Time	Temporary	Chronic-Not Severe	Chronic- Severe
Benefit Receipt Rate					
Social Security	0.173	0.089	0.132	0.184	0.348
Supplemental Security Income	0.051	0.010	0.028	0.056	0.140
Social Security or SSI	0.203	0.095	0.149	0.219	0.433
Workers' Compensation	0.030	0.021	0.039	0.030	0.030
Unemployment Insurance	0.065	0.083	0.069	0.062	0.035
Food Stamps	0.226	0.125	0.178	0.251	0.425
Subsidized Housing	0.070	0.049	0.047	0.081	0.117
Any one of the above	0.441	0.298	0.369	0.465	0.744
Work and Wealth					
Not receiving any benefit above and working fewer than 1000 hours	0.206	0.247	0.190	0.200	0.177
Median Net Wealth at Onset (2016 Dollars)	\$39,786	\$51,330	\$39,636	\$34,175	\$17,508
Median Net Wealth (2016 Dollars)	\$50,284	\$65,590	\$49,996	\$42,653	\$29,462

Notes: Unless indicated otherwise, the benefit receipt rates and median net wealth are numbers for the 6-10 years after a woman's year of disability onset. Benefits received are reported at the family level. The receipt rates are not adjusted for underreporting.

Year from	All Public	AFDC/TANF	SNAP/Food	Unemployment	Workers'	Supplemental	Social Security (OASI and	Federal Income
Onset	Transfers		Stamps	Insurance	Compensation	Security Income	(ONDF and SSDI)	Tax
-5	671**	95	19	151	-108	-8	208	-317
	(254)	(87)	(23)	(135)	(103)	(36)	(117)	(493)
-4	850**	166	10	181	-50	39	396**	-144
	(306)	(129)	(19)	(174)	(120)	(44)	(123)	(593)
-3	569*	124	-3	64	-42	-30	431**	-841
	(237)	(110)	(22)	(62)	(103)	(43)	(133)	(563)
-2	1,056**	240	71**	97	-29	7	596**	-125
	(278)	(137)	(22)	(75)	(122)	(49)	(147)	(644)
-1	1,464**	254*	96**	144	122	44	625**	-590
	(273)	(116)	(26)	(78)	(121)	(54)	(154)	(762)
0	1,566**	287*	119**	161	63	52	712**	-336
	(302)	(146)	(25)	(93)	(131)	(56)	(152)	(1,001)
1	2,830**	333**	165**	212*	658**	169**	967**	-1,578*
	(322)	(123)	(29)	(98)	(187)	(66)	(168)	(796)
2	2,891**	458**	171**	234*	521	83	1,120**	-1,676*
	(528)	(150)	(27)	(102)	(368)	(56)	(176)	(719)
3	2,654**	115	207**	88	286	242**	1,380**	-2,117**
	(324)	(129)	(32)	(80)	(152)	(66)	(191)	(716)
4	2,293**	-4	167**	139	119	267**	1,332**	-1627
	(335)	(141)	(27)	(105)	(153)	(67)	(191)	(866)
5	2,711**	-53	152**	156	256	397**	1,516**	-2,349**
	(342)	(133)	(28)	(98)	(169)	(82)	(186)	(899)
6	2,459**	-37	161**	105	405*	340**	1,206**	-1105
	(357)	(159)	(28)	(112)	(196)	(66)	(187)	(1,064)
7	2,331**	-143	211**	96	50	374**	1,498**	-840
	(333)	(138)	(41)	(89)	(140)	(77)	(201)	(1,243)
8	2,407**	-170	174**	224*	460*	430**	1,246**	-2,172**
	(363)	(166)	(30)	(105)	(196)	(80)	(194)	(807)
9	2,467**	-251	192**	199	357	437**	1,496**	-2,062*
	(444)	(151)	(35)	(140)	(293)	(81)	(213)	(818)
10	2,783**	-558**	214**	789	221	599**	1,478**	-2,354*
	(865)	(176)	(36)	(756)	(175)	(95)	(210)	(1,070)

Table 7: Changes in Amount of Public Transfers received relative to Baseline, All-Disabled Women

Year from Onset	All Public Transfers	AFDC/TANF	SNAP/Food Stamps	Unemployment Insurance	Workers' Compensation	Supplemental Security Income	Social Security (OASI and SSDI)	Federal Income Tax
-5	822	-372	80	-236	83	-60	975*	-1,222*
	(705)	(252)	(84)	(139)	(411)	(119)	(409)	(579)
-4	1274	-98	-30	-461**	16	332	1,804**	-2,031**
	(975)	(434)	(60)	(106)	(332)	(265)	(502)	(578)
-3	804	-27	-98	-205	-66	96	1,290**	-2,939**
	(831)	(342)	(62)	(160)	(270)	(246)	(439)	(750)
-2	1,209	-345	3	-325*	-309	9	2,400**	-1,998**
	(1,049)	(544)	(66)	(131)	(358)	(191)	(583)	(682)
-1	2,391**	256	83	-195	298	119	1,834**	-3,135**
	(904)	(408)	(76)	(167)	(439)	(238)	(478)	(768)
0	2,716**	-264	138	129	159	38	2,519**	-3,199**
	(994)	(542)	(73)	(251)	(409)	(205)	(547)	(721)
1	5,587**	53	322**	22	956	263	3,565**	-4,088**
	(930)	(384)	(98)	(205)	(518)	(238)	(591)	(736)
2	5,836**	59	261**	-145	320	321	3,970**	-3,956**
	(988)	(478)	(81)	(208)	(461)	(206)	(597)	(700)
3	7,550**	-231	249**	22	1001	752**	4,822**	-4,081**
	(1,051)	(390)	(86)	(221)	(600)	(278)	(701)	(928)
4	6,879**	-411	316**	-245	421	903**	4,943**	-3,962**
	(1,039)	(463)	(80)	(143)	(504)	(246)	(677)	(734)
5	8,063**	-364	302**	-169	923	1,273**	5,679**	-4,998**
	(1,108)	(481)	(83)	(155)	(684)	(288)	(656)	(822)
6	9,182**	329	411**	-185	1209	1,456**	5,049**	-4,533**
	(1,216)	(550)	(94)	(185)	(807)	(304)	(648)	(904)
7	9,002**	132	444**	-273	676	1,727**	5,902**	-4,516**
	(1,152)	(524)	(97)	(158)	(670)	(374)	(706)	(1,010)
8	8,802**	57	403**	-116	1214	1,800**	5,144**	-5,463**
	(1,188)	(591)	(88)	(165)	(753)	(383)	(657)	(879)
9	7,759**	-953	476**	-225	443	2,036**	5,558**	-5,168**
	(1,195)	(540)	(97)	(180)	(680)	(380)	(706)	(953)
10	7,727**	-1,430*	478**	-316	798	2,593**	5,578**	-6,374**
	(1,270)	(580)	(91)	(183)	(737)	(405)	(718)	(905)

 Table 8: Changes in Amount of Public Transfers (in 2016 Dollars) received relative to Baseline, Chronic-Severe Disabled Women



Figure 1 Lifetime Disability Prevalence Rates for Men and Women

Figure 2 Change in Annual Hours of Work relative to Baseline, Disabled Women





Figure 3

Figure 4 Change in Annual Earnings relative to Baseline, Disabled Women





Figure 5 Change in After-tax Pre-Transfer Income relative to Baseline, Disabled Women

Figure 6 Change in After-tax Post-Transfer Income relative to Baseline, Disabled Women





Figure 7 Change in Public Transfer Income (in 2016 dollars) relative to Baseline, Disabled Women

Figure 8 Poverty Rate Before and After Disability Onset, Disabled Women





Figure 10 Change in Food plus Housing Consumption relative to Baseline, Disabled Women





Figure 11 Consumption Poverty Rate Before and After Disability Onset, Disabled Women

Appendices

A. Controls for individual fixed effects regressions

All regressions include individual fixed effects and the time from onset dummies, year dummies, state dummies, age, age-squared, a married indicator, number of children, education dummies (12 yrs, 13-15 yrs, 16, 17+), also head status interacted with age, and age-squared. The following co-variates are also included, depending on the outcome variable:

Outcome	Additional Controls
Annual Earnings,	Education*Age interactions, Education*Age-sq interactions,
Hours, Hourly	Education*(linear time trend) interactions, Education*(linear time
Earnings, Taxes	trend)-squared interactions
Income	Education*Age interactions, Education*Age-sq interactions,
	Education*(linear time trend) interactions, Education*(linear time
	trend)-squared interactions, number of family members, married
	indicator*(husband disability) interaction, married*(husband's age)
	interaction
Food, Food at Home,	Number of men, number of women, number of kids (ages <=10),
Food Away from	number of young adults (ages 11-17), number of elders (ages >=65)
Home, Food plus	- And the squared of these five variables.
Housing	
	married indicator*(husband current disability indicator) interaction,
	married*(husband's age) interaction
Public Transfers, and	Number of family members
various benefits	

B. Additional Figures for various economic outcomes



Figure A2 Change in After-tax Income with non-SSA Transfers relative to Baseline, Disabled Women





Figure A4 Change in AFDC/TANF Received (in 2016 dollars) relative to Baseline, Disabled Women





Figure A5 Change in Food Stamps/SNAP Received (in 2016 dollars) relative to Baseline, Disabled Women

Figure A6 Change in Unemployment Insurance Received (in 2016 dollars) relative to Baseline, Disabled Women





Figure A7 Change in Workers' Compensation Received (in 2016 dollars) Before and After Disability Onset Extent of Disability Groups and All Disabled

Figure A8 Change in Supplemental Security Income (in 2016 dollars) relative to Baseline, Disabled Women





Figure A9 Change in Social Security Income (in 2016 dollars) relative to Baseline, Disabled Women

Figure A10 Change in estimated Federal Income Tax payable (in 2016 dollars) relative to Baseline, Disabled Women





Figure A11 Change in Annual Hours of Work relative to Baseline, Groups Defined by SSA Benefit Receipt and Disability

Figure A12 Change in After-Tax Post-Transfer Income relative to Baseline, Groups Defined by SSA Benefit Receipt and Disability



Figure A13 Change in Food plus Housing Consumption relative to Baseline Groups Defined by SSA Benefit Receipt and Disability



Appendix Table 1: Changes in Economic Outcomes Relative to Baseline, All Disabled Male Family Heads

Year from Onset	Hours	Fraction not working	Earnings	Pre-tax Pre- transfer Income	After-tax Income without Transfers	After-tax income with non-SSA Transfers	After-tax income with SSA Transfers	After Tax Income with all Transfers	Food Consumption	Housing Consumption	Food plus Housing Consumption
-5	-2	0.03	-5.41**	-2.72	-2.22	-2.26	-2.07	-1.99	0.57	-4.18**	-2.37*
	(20)		(1.93)	(1.65)	(1.30)	(1)	(1.51)	(1.28)	(1.38)	(1.48)	(1.11)
-4	-38	0.03	-4.57	-2.51	-2.27	-3.40*	-2.75	-1.94	-2.28*	-3.93*	-3.83**
	(23)		(2.55)	(1.93)	(1.48)	(2)	(1.62)	(1.49)	(1.15)	(1.66)	(1.17)
-3	-47*	0.04	-6.98**	-4.54*	-3.90*	-5.39**	-5.05**	-3.82*	-2.38	-4.55**	-4.25**
	(22)		(2.38)	(2.27)	(1.68)	(2)	(1.81)	(1.66)	(1.28)	(1.75)	(1.24)
-2	-95**	0.03	-9.70**	-6.56**	-5.77**	-7.00**	-6.71**	-5.52**	-0.65	-4.45*	-3.53**
	(25)		(1.94)	(1.70)	(1.43)	(2)	(1.53)	(1.38)	(1.29)	(1.90)	(1.33)
-1	-173**	0.06	-12.68**	-8.13**	-7.09**	-7.28**	-7.81**	-5.19**	-2.81*	-2.76	-3.65*
	(25)		(2.03)	(1.83)	(1.53)	(2)	(1.76)	(1.53)	(1.28)	(2.19)	(1.43)
0	-273**	0.06	-15.78**	-11.39**	-10.34**	-8.39**	-10.37**	-6.34**	-2.90*	-6.44**	-5.78**
	(27)		(2.06)	(1.79)	(1.50)	(2)	(1.60)	(1.50)	(1.23)	(2.00)	(1.39)
1	-398**	0.14	-22.53**	-16.13**	-15.22**	-12.26**	-14.22**	-9.16**	-2.51	-5.09*	-4.89**
	(28)		(2.24)	(1.90)	(1.57)	(2)	(1.74)	(1.61)	(1.52)	(2.09)	(1.51)
2	-377**	0.14	-22.38**	-14.91**	-14.31**	-12.28**	-12.93**	-9.49**	-6.16**	-6.83**	-7.83**
	(29)		(2.24)	(2.05)	(1.61)	(2)	(1.74)	(1.65)	(1.27)	(2.15)	(1.48)
3	-360**	0.17	-21.28**	-14.60**	-13.86**	-13.54**	-12.72**	-9.93**	-5.31**	-6.20**	-7.18**
	(29)		(2.66)	(2.53)	(2.00)	(2)	(2.31)	(2.07)	(1.39)	(2.35)	(1.61)
4	-400**	0.15	-23.09**	-14.63**	-13.17**	-11.37**	-10.54**	-8.03**	-6.26**	-6.99**	-8.05**
	(30)		(2.41)	(2.42)	(2.18)	(2)	(2.42)	(2.25)	(1.35)	(2.47)	(1.68)
5	-384**	0.19	-24.16**	-17.49**	-15.79**	-15.47**	-13.50**	-10.45**	-6.63**	-5.94*	-7.97**
	(30)		(2.81)	(2.34)	(1.89)	(2)	(2.13)	(1.91)	(1.39)	(2.73)	(1.82)
6	-366**	0.16	-22.44**	-15.47**	-14.39**	-13.97**	-12.43**	-10.22**	-4.87**	-6.57*	-7.38**
	(32)		(2.70)	(2.41)	(1.97)	(2)	(2.17)	(1.99)	(1.40)	(3.29)	(2.15)
7	-371**	0.20	-24.81**	-15.14**	-13.34**	-13.13**	-10.84**	-8.62**	-7.86**	-10.32**	-11.06**
	(32)		(2.90)	(2.90)	(2.69)	(3)	(2.91)	(2.74)	(1.39)	(2.60)	(1.79)
8	-338**	0.16	-24.86**	-17.87**	-16.05**	-16.55**	-14.15**	-12.22**	-4.70**	-9.14**	-9.16**
	(33)		(2.72)	(2.48)	(2.08)	(2)	(2.28)	(2.06)	(1.48)	(2.98)	(2.02)
9	-373**	0.20	-23.61**	-17.17**	-15.84**	-16.07**	-14.20**	-11.40**	-6.87**	-8.78**	-10.02**
	(34)		(3.01)	(2.60)	(2.14)	(2)	(2.35)	(2.14)	(1.51)	(3.16)	(2.09)
10	-413**	0.18	-24.63**	-16.28**	-15.08**	-15.06**	-12.63**	-9.29**	-6.47**	-9.68**	-10.18**
	(37)		(3.19)	(2.89)	(2.35)	(3)	(2.56)	(2.44)	(1.59)	(3.01)	(2.10)

Year from Onset	All Public Transfers	AFDC/TANF	SNAP/Food Stamps	Unemployment Insurance	Workers' Compensation	Supplemental Security Income	Social Security (OASI and SSDI)	Federal Income Tax
-5	249	14	17	55	64	13	143**	-1,108**
	(179)	(20)	(14)	(40)	(60)	(18)	(47)	(394)
-4	319	-2	18	69	88	44	219**	-1,303**
	(210)	(20)	(16)	(43)	(56)	(25)	(64)	(409)
-3	472*	32	20	97*	40	31	242**	-1,830**
	(210)	(24)	(16)	(44)	(54)	(27)	(64)	(541)
-2	604**	19	58**	126**	184**	89**	344**	-1,884**
	(224)	(26)	(19)	(45)	(62)	(30)	(72)	(422)
-1	1,330**	16	68**	65	461**	97**	370**	-2,484**
	(248)	(22)	(20)	(39)	(103)	(35)	(77)	(521)
0	2,973**	16	126**	84*	918**	171*	474**	-2,447**
	(319)	(26)	(23)	(41)	(137)	(82)	(72)	(491)
1	4,360**	68**	129**	50	1,144**	164**	733**	-2,775**
	(368)	(26)	(22)	(43)	(167)	(42)	(100)	(642)
2	3,910**	54	113**	47	748**	161**	804**	-2,786**
	(376)	(29)	(23)	(43)	(132)	(40)	(94)	(669)
3	3,514**	-2	109**	43	582**	222**	971**	-3,259**
	(464)	(25)	(23)	(50)	(182)	(45)	(109)	(693)
4	3,750**	12	127**	84	534**	242**	1,033**	-3,384**
	(330)	(27)	(23)	(48)	(101)	(45)	(103)	(634)
5	3,821**	17	101**	29	545**	304**	1,329**	-4,106**
	(334)	(21)	(23)	(49)	(115)	(65)	(130)	(744)
6	3,480**	40	128**	63	507**	219**	1,158**	-3,691**
	(361)	(28)	(23)	(62)	(111)	(42)	(112)	(726)
7	3,684**	36	129**	50	481**	243**	1,398**	-4,151**
	(341)	(25)	(25)	(62)	(127)	(54)	(131)	(806)
8	3,490**	37	106**	-43	484**	269**	1,248**	-4,279**
	(353)	(31)	(25)	(43)	(108)	(55)	(120)	(789)
9	3,950**	41	107**	146	479**	263**	1,411**	-4,259**
	(407)	(28)	(25)	(76)	(131)	(53)	(133)	(880)
10	3,854**	22	94**	2	554**	287**	1,429**	-4,272**
	(393)	(27)	(22)	(51)	(123)	(55)	(134)	(920)

Appendix Table 2: Changes in Amount of Public Transfers Received Relative to Baseline, All Disabled Male Family Heads

		Fraction		Pre-tax Pre-	After-tax	After-tax income	After-tax	After Tax	Food	Hausing	Food plus
Year from		not		transfer	Income without	with non-SSA	income with	Income with	Congumation	Consumption	Housing
Onset	Hours	working	Earnings	Income	Transfers	Transfers	SSA Transfers	all Transfers	Consumption	Consumption	Consumption
-5	-57	0.04	-11.55**	-9.25**	-8.13**	-6.85*	-7.38*	-6.79*	-0.16	-9.78**	-5.69**
	(57)		(3.72)	(3.18)	(2.90)	(3)	(3.14)	(2.67)	(2.63)	(2.84)	(2.10)
-4	-144**	0.03	-12.49**	-7.74*	-6.44*	-10.72**	-6.82*	-8.23**	-5.03	-7.45*	-8.10**
	(51)		(3.54)	(3.33)	(3.13)	(3)	(3.22)	(2.90)	(2.60)	(3.16)	(2.29)
-3	-102	0.04	-14.29**	-8.89*	-7.29*	-7.70*	-7.56*	-7.04*	-5.91*	-5.28	-7.93**
	(52)		(3.72)	(3.61)	(3.24)	(3)	(3.33)	(2.93)	(3.01)	(4.14)	(2.61)
-2	-117	0.04	-16.12**	-11.03**	-9.67**	-10.38**	-9.83**	-10.15**	-7.98**	-10.72**	-11.73**
	(62)		(4.45)	(3.78)	(3.42)	(4)	(3.58)	(3.01)	(2.90)	(4.13)	(2.86)
-1	-346**	0.08	-24.43**	-14.74**	-12.42**	-11.26**	-11.29**	-9.62**	-9.74**	-11.73**	-12.80**
	(61)		(4.45)	(3.76)	(3.43)	(4)	(3.82)	(3.25)	(3.17)	(4.19)	(2.85)
0	-720**	0.16	-38.83**	-25.00**	-23.49**	-16.13**	-19.49**	-12.49**	-9.28**	-12.82**	-13.20**
	(71)		(4.70)	(4.32)	(3.81)	(4)	(3.78)	(3.41)	(2.76)	(4.09)	(2.69)
1	-1,161**	0.42	-60.91**	-40.47**	-38.15**	-26.17**	-30.51**	-18.48**	-8.25	-16.85**	-14.87**
	(68)		(3.95)	(3.72)	(3.41)	(4)	(3.75)	(3.48)	(4.45)	(4.12)	(3.25)
2	-1,231**	0.47	-63.97**	-43.32**	-40.97**	-34.92**	-29.72**	-23.75**	-16.92**	-18.92**	-20.93**
	(72)		(3.94)	(3.60)	(3.37)	(3)	(3.38)	(3.15)	(2.65)	(3.67)	(2.50)
3	-1,328**	0.58	-67.47**	-45.95**	-44.78**	-35.27**	-32.31**	-24.17**	-15.62**	-18.34**	-19.72**
	(65)		(3.99)	(3.93)	(3.50)	(5)	(3.85)	(3.88)	(3.20)	(3.67)	(2.65)
4	-1,365**	0.55	-68.38**	-46.42**	-43.26**	-33.05**	-25.47**	-19.42**	-18.83**	-18.09**	-21.70**
	(69)		(4.00)	(4.98)	(5.24)	(7)	(7.26)	(5.09)	(2.55)	(4.09)	(2.63)
5	-1,430**	0.64	-74.98**	-50.71**	-48.62**	-42.14**	-32.45**	-25.20**	-18.92**	-18.50**	-22.21**
	(64)		(3.26)	(3.60)	(3.30)	(4)	(3.55)	(3.22)	(2.84)	(4.06)	(2.68)
6	-1,474**	0.60	-74.72**	-51.57**	-48.77**	-40.17**	-32.92**	-25.45**	-14.82**	-21.59**	-22.02**
	(71)		(3.61)	(3.72)	(3.56)	(4)	(3.78)	(3.69)	(3.09)	(4.56)	(2.93)
7	-1,460**	0.68	-79.99**	-53.27**	-50.43**	-44.98**	-30.52**	-27.31**	-19.33**	-24.66**	-25.96**
	(70)		(2.98)	(3.73)	(3.64)	(4)	(4.20)	(3.58)	(3.14)	(5.36)	(3.01)
8	-1,542**	0.65	-80.67**	-59.47**	-56.68**	-50.73**	-40.04**	-34.62**	-19.47**	-28.57**	-28.27**
	(73)		(3.52)	(3.43)	(3.29)	(3)	(3.61)	(3.27)	(2.58)	(4.68)	(2.71)
9	-1,493**	0.72	-79.72**	-58.72**	-56.53**	-50.79**	-36.86**	-30.50**	-20.62**	-20.65**	-25.08**
	(80)		(3.60)	(3.56)	(3.39)	(4)	(3.60)	(3.34)	(2.75)	(7.06)	(3.76)
10	-1,487**	0.66	-77.23**	-55.39**	-53.09**	-49.66**	-34.84**	-28.32**	-16.18**	-24.70**	-25.27**
	(83)		(4.39)	(4.66)	(4.30)	(4)	(4.35)	(4.34)	(3.49)	(4.46)	(2.92)

Appendix Table 3: Changes in Economic Outcomes Relative to Baseline, Chronic-Severe Disabled Male Family Heads

Year from Onset	All Public Transfers	AFDC/TANF	SNAP/Food Stamps	Unemployment Insurance	Workers' Compensation	Supplemental Security Income	Social Security (OASI and SSDI)	Federal Income Tax
-5	1,352**	9	47	82	253	42	555**	-1,815**
	(512)	(59)	(40)	(75)	(150)	(42)	(178)	(432)
-4	65	-43	33	-68	474*	63	783**	-2,120**
	(501)	(48)	(52)	(39)	(212)	(61)	(269)	(389)
-3	1,190*	28	49	101	352	60	787**	-2,732**
	(600)	(87)	(55)	(87)	(183)	(54)	(247)	(525)
-2	1,370*	126	75	200	615*	206*	1,036**	-2,819**
	(687)	(105)	(59)	(151)	(239)	(99)	(237)	(483)
-1	2,656**	-43	106*	113	956**	206*	1,257**	-3,740**
	(800)	(64)	(53)	(87)	(359)	(87)	(302)	(614)
0	7,309**	-19	208**	188	1,839**	293**	1,591**	-3,613**
	(1,201)	(70)	(63)	(101)	(418)	(102)	(254)	(592)
1	12,033**	69	279**	-40	2,594**	545**	2,518**	-5,092**
	(1,207)	(73)	(60)	(66)	(554)	(152)	(362)	(686)
2	10,956**	96	237**	-59	1,500**	670**	3,253**	-4,877**
	(1,078)	(88)	(64)	(55)	(414)	(176)	(360)	(605)
3	13,560**	-3	214**	93	2,459*	782**	3,775**	-5,300**
	(2,204)	(82)	(52)	(157)	(987)	(185)	(397)	(755)
4	13,690**	46	349**	-103	1,830**	776**	4,240**	-5,351**
	(1,171)	(67)	(68)	(62)	(466)	(177)	(368)	(670)
5	13,871**	108	260**	-176**	1,545**	904**	5,035**	-6,267**
	(1,063)	(72)	(65)	(57)	(396)	(194)	(447)	(753)
6	14,874**	33	343**	77	1,800**	896**	4,811**	-6,263**
	(1,264)	(72)	(76)	(240)	(464)	(183)	(392)	(709)
7	14,432**	111	351**	-182**	1,542**	1,017**	5,708**	-6,955**
	(1,173)	(70)	(78)	(70)	(489)	(242)	(458)	(799)
8	13,953**	51	314**	-213**	1,805**	1,019**	5,225**	-7,116**
	(1,323)	(91)	(101)	(56)	(514)	(249)	(430)	(778)
9	14,814**	144	348**	-117	1,825**	1,121**	6,187**	-6,984**
	(1,285)	(103)	(108)	(98)	(557)	(237)	(475)	(784)
10	15,201**	263*	327**	-110	1,598**	1,072**	6,089**	-7,508**
	(1,405)	(116)	(78)	(121)	(433)	(242)	(504)	(883)

Appendix Table 4: Changes in Amount of Public Transfers Received Relative to Baseline, Chronic-Severe Disabled Male Family Heads

Year from Onset	Hours	Fraction not working	Earnings	Pre-tax Pre- transfer Income	After-tax Income without Transfers	After-tax income with non-SSA Transfers	After-tax income with SSA Transfers	After Tax Income with all Transfers	Food Consumption	Housing Consumption	Food plus Housing Consumption
-5	28	0.31	1.02	-3.48	-2.54	-2.02	-2.76	-2.35	0.1	-1.57	-0.94
	(28)		(3.02)	(1.91)	(1.62)	(1)	(1.55)	(1.42)	(1.42)	(2.47)	(1.49)
-4	56	0.28	3.32	-2.49	-1.7	-1.04	-1.67	-1.15	-0.64	-3.73*	-2.36
	(29)		(3.57)	(2.22)	(1.86)	(2)	(1.78)	(1.69)	(1.45)	(1.89)	(1.29)
-3	-7	0.28	-0.44	-4.16	-3.13	-2.46	-2.99	-2.5	-0.85	-2.44	-2.02
	(32)		(3.70)	(2.29)	(1.95)	(2)	(1.87)	(1.78)	(1.55)	(2.61)	(1.63)
-2	36	0.27	4	-2.37	-2.43	-1.3	-2.22	-1.21	-2.86	-5.29*	-4.31**
	(34)		(4.02)	(2.57)	(2.14)	(2)	(2.06)	(1.94)	(1.51)	(2.35)	(1.56)
-1	-9	0.27	2.25	-3.86	-3.79	-1.96	-3.41	-1.68	-0.77	-2.39	-2.06
	(35)		(4.15)	(2.40)	(2.02)	(2)	(1.94)	(1.83)	(1.53)	(2.53)	(1.67)
0	-106**	0.28	-9.22*	-7.68**	-7.06**	-5.13*	-7.00**	-5.07**	0.12	-4.96	-2.89
	(37)		(3.97)	(2.47)	(2.10)	(2)	(2.02)	(1.90)	(2.25)	(2.73)	(1.96)
1	-222**	0.33	-16.38**	-9.63**	-8.80**	-4.98*	-8.39**	-4.53*	-0.9	-6.07*	-4.22*
	(36)		(3.92)	(2.48)	(2.19)	(2)	(2.27)	(2.14)	(2.22)	(2.74)	(1.92)
2	-179**	0.33	-13.70**	-12.37**	-11.48**	-7.00**	-9.49**	-6.36**	-1.67	-4.62	-3.86*
	(37)		(3.95)	(2.36)	(2.05)	(2)	(2.06)	(1.94)	(1.71)	(2.91)	(1.92)
3	-183**	0.34	-15.70**	-14.20**	-13.14**	-9.03**	-10.99**	-7.89**	-2.74	-8.03*	-6.12**
	(39)		(3.93)	(2.33)	(2.07)	(2)	(2.04)	(1.96)	(1.76)	(3.14)	(2.04)
4	-195**	0.36	-18.52**	-13.46**	-12.19**	-11.02**	-11.68**	-9.48**	-4.67**	-6.11*	-6.09**
	(40)		(3.96)	(2.51)	(2.24)	(2)	(2.03)	(1.95)	(1.77)	(3.09)	(2.04)
5	-233**	0.37	-20.11**	-15.34**	-14.11**	-12.58**	-12.67**	-10.54**	-3.62	-7.55*	-6.64**
	(40)		(4.09)	(2.45)	(2.19)	(2)	(2.24)	(2.10)	(1.91)	(3.07)	(2.06)
6	-215**	0.38	-19.17**	-14.29**	-13.21**	-10.07**	-10.17**	-8.24**	-4.44*	-9.21**	-7.81**
	(41)		(4.04)	(2.58)	(2.31)	(2)	(2.36)	(2.24)	(1.80)	(3.55)	(2.29)
7	-246**	0.39	-18.85**	-11.52**	-11.32**	-9.49**	-8.48**	-7.84**	-2	-9.42*	-7.12**
	(41)		(4.20)	(3.64)	(2.91)	(3)	(2.99)	(2.79)	(2.63)	(3.96)	(2.60)
8	-232**	0.37	-18.42**	-12.42**	-11.44**	-10.51**	-9.85**	-8.85**	-3.80*	-8.38*	-7.59**
	(43)		(4.35)	(2.72)	(2.41)	(2)	(2.32)	(2.21)	(1.81)	(3.64)	(2.35)
9	-251**	0.39	-20.88**	-13.12**	-11.43**	-10.53**	-9.41**	-8.61**	-4.14*	-7.89*	-6.96**
	(44)		(4.30)	(2.97)	(2.74)	(3)	(2.64)	(2.56)	(1.94)	(3.57)	(2.38)
10	-240**	0.38	-20.73**	-11.86**	-11.34**	-10.06**	-8.74**	-7.74*	-3.01	-4.99	-4.97
	(46)		(4.56)	(4.34)	(3.42)	(4)	(3.33)	(3.50)	(2.19)	(5.06)	(3.20)

Appendix Table 5: Changes in Economic Outcomes Relative to Baseline, All-Disabled Women with Year of Onset prior to 1990

Year from	All Public	AFDC/TANF	SNAP/Food	Unemployment	Workers'	Supplemental	Social Security (OASI and	Federal Income
Onset	Transfers		Stamps	Insurance	Compensation	Security Income	SSDI)	Tax
-5	283	189	14	14	-197*	-19	-39	-659
	(277)	(145)	(20)	(77)	(82)	(49)	(141)	(436)
-4	509	284	-7	359	-160	16	86	-545
	(405)	(189)	(21)	(258)	(87)	(62)	(155)	(524)
-3	483	282	8	220*	-77	8	146	-948
	(349)	(193)	(23)	(100)	(103)	(67)	(174)	(535)
-2	946*	356	46	287*	36	6	210	-318
	(375)	(199)	(26)	(116)	(129)	(62)	(188)	(607)
-1	1,351**	460*	46	386**	59	5	310	-542
	(392)	(191)	(25)	(131)	(135)	(69)	(203)	(595)
0	1,279**	421*	68*	313	116	-44	311	-1,251*
	(419)	(211)	(29)	(161)	(139)	(70)	(197)	(625)
1	2,397**	636**	129**	464**	409*	5	402*	-1,710**
	(452)	(218)	(30)	(177)	(191)	(79)	(203)	(633)
2	2,069**	721**	140**	370**	148	-25	530*	-1,853**
	(415)	(223)	(29)	(143)	(151)	(58)	(216)	(606)
3	2,192**	252	143**	334*	406*	116	639**	-2,124**
	(453)	(206)	(28)	(142)	(206)	(76)	(230)	(597)
4	2,082**	107	155**	371*	214	226*	716**	-2,619**
	(464)	(205)	(29)	(180)	(203)	(89)	(232)	(582)
5	2,367**	-41	151**	391*	428	347**	844**	-2,585**
	(491)	(219)	(29)	(198)	(248)	(99)	(235)	(618)
6	2,426**	86	163**	301	572*	334**	734**	-2,222**
	(498)	(219)	(30)	(209)	(259)	(83)	(230)	(646)
7	1,893**	-44	162**	370*	152	318**	755**	-1595
	(458)	(210)	(32)	(178)	(185)	(84)	(230)	(928)
8	2,025**	-80	128**	398*	494*	415**	705**	-2,360**
	(471)	(218)	(31)	(186)	(214)	(92)	(233)	(673)
9	2,141**	-207	166**	391	607	399**	910**	-2,890**
	(634)	(224)	(33)	(252)	(422)	(90)	(247)	(693)
10	2,749*	-527*	184**	1148	355	608**	1,007**	-2090
	(1,146)	(229)	(37)	(1,003)	(216)	(111)	(249)	(1,072)

Appendix Table 6: Changes in Amount of Public Transfers Received Relative to Baseline, All-Disabled Women with Year of Onset prior to 1990

Year		Fraction		Pre-tax Pre-	After-tax	After-tax income	After-tax	After Tax	Food	Hausina	Food plus
from		not		transfer	Income without	with non-SSA	income with	Income with	Consumption	Consumption	Housing
Onset	Hours	working	Earnings	Income	Transfers	Transfers	SSA Transfers	all Transfers	Consumption	Consumption	Consumption
-5	3	0.31	-0.50	-2.34	-2.16	-3	0.61	-1.35	0.62	2.42	0.57
	(64)		(7.85)	(5.29)	(4.97)	(4)	(4.62)	(3.40)	(5.18)	(5.99)	(4.08)
-4	39	0.30	0.84	-7.96	-5.95	-6	-0.78	-1.91	-4.88	-1.33	-4.45
	(79)		(10.81)	(5.32)	(5.10)	(5)	(4.29)	(3.89)	(4.64)	(5.10)	(3.49)
-3	-100	0.28	-9.37	-6.62	-3.14	-3	1.59	0.39	-0.08	-7.22	-4.38
	(83)		(10.31)	(5.39)	(5.00)	(5)	(4.51)	(4.26)	(4.89)	(5.12)	(3.75)
-2	-74	0.28	-14.61	-6.41	-4.03	-7	1.87	-1.93	-0.02	-9.23	-5.70
	(90)		(10.12)	(5.94)	(5.57)	(5)	(4.94)	(4.56)	(5.97)	(5.59)	(4.11)
-1	-186*	0.33	-21.63*	-16.11**	-13.50**	-11.39*	-8.09	-7.37	3.68	-0.30	-0.36
	(83)		(10.14)	(5.51)	(5.21)	(5)	(4.74)	(4.38)	(5.25)	(6.93)	(4.28)
0	-322**	0.37	-38.79**	-18.89**	-17.26**	-13.85**	-12.70*	-10.59*	-1.17	-3.64	-4.66
	(94)		(8.31)	(5.96)	(5.50)	(5)	(5.38)	(4.92)	(5.31)	(7.06)	(4.57)
1	-664**	0.52	-59.56**	-27.20**	-24.37**	-15.45**	-17.43**	-10.22*	0.48	-5.62	-4.65
	(88)		(6.18)	(5.19)	(4.96)	(5)	(4.97)	(4.53)	(5.58)	(6.67)	(4.33)
2	-760**	0.59	-65.15**	-37.72**	-35.37**	-21.93**	-24.33**	-15.19**	-1.94	-4.44	-5.03
	(94)		(6.19)	(5.13)	(4.92)	(5)	(5.04)	(4.47)	(5.61)	(6.94)	(4.58)
3	-777**	0.63	-67.71**	-37.30**	-35.30**	-24.41**	-24.28**	-15.55**	-7.96	-5.47	-8.13
	(92)		(5.75)	(4.32)	(4.19)	(4)	(4.32)	(4.08)	(5.39)	(8.58)	(5.03)
4	-827**	0.67	-68.54**	-34.00**	-31.77**	-24.60**	-19.04**	-14.09**	-4.68	-10.07	-9.45*
	(96)		(5.98)	(5.01)	(4.85)	(4)	(4.83)	(4.22)	(5.39)	(7.14)	(4.33)
5	-858**	0.66	-69.82**	-37.62**	-35.97**	-28.74**	-20.20**	-16.27**	-7.11	-7.37	-10.08*
	(96)		(6.14)	(5.79)	(5.46)	(5)	(5.73)	(5.02)	(5.14)	(8.22)	(5.07)
6	-882**	0.72	-73.21**	-32.31**	-30.98**	-19.49**	-15.31*	-6.94	0.55	-11.96	-7.61
	(98)		(5.95)	(7.18)	(6.68)	(7)	(7.19)	(6.13)	(5.90)	(6.60)	(4.82)
7	-910**	0.73	-66.83**	-37.74**	-37.35**	-26.91**	-18.33**	-13.19*	-6.09	-1.75	-6.37
	(94)		(9.24)	(6.50)	(5.72)	(5)	(6.24)	(5.44)	(5.46)	(15.55)	(7.97)
8	-910**	0.72	-71.76**	-39.07**	-36.20**	-25.71**	-19.53**	-13.02*	-2.96	-9.92	-9.14*
	(97)		(6.06)	(5.60)	(5.45)	(6)	(5.64)	(5.52)	(5.27)	(7.41)	(4.65)
9	-971**	0.79	-79.59**	-44.29**	-41.40**	-33.87**	-21.67**	-19.43**	-1.68	-10.08	-8.22
	(97)		(4.95)	(5.86)	(5.56)	(5)	(5.95)	(5.06)	(6.57)	(7.21)	(5.02)
10	-943**	0.77	-76.57**	-40.63**	-37.36**	-33.76**	-16.34**	-18.21**	-6.95	-13.00	-12.44**
	(99)		(5.16)	(5.08)	(5.05)	(5)	(5.56)	(4.70)	(5.70)	(7.19)	(4.78)

Appendix Table 7: Changes in Economic Outcomes Relative to Baseline, Chronic-Severe Disabled Women with Year of Onset prior to 1990

Year from Onset	All Public Transfers	AFDC/TANF	SNAP/Food Stamps	Unemployment Insurance	Workers' Compensation	Supplemental Security Income	Social Security (OASI and SSDI)	Federal Income Tax
-5	656	-826	74	-280	-448	-90	1,351*	-639
	(845)	(423)	(68)	(156)	(390)	(131)	(625)	(550)
-4	1073	-280	11	-331*	-344	480	1,709*	-1,722**
	(1,400)	(582)	(74)	(150)	(422)	(372)	(714)	(615)
-3	1670	-27	70	-48	-201	392	1,514*	-2,106**
	(1,402)	(556)	(84)	(185)	(445)	(391)	(706)	(805)
-2	1,643	-443	40	-121	-138	142	2,143**	-1,775*
	(1,442)	(767)	(90)	(188)	(479)	(274)	(784)	(824)
-1	2417	200	110	-32	-110	153	1,860*	-2,300**
	(1,334)	(692)	(85)	(213)	(502)	(300)	(723)	(798)
0	2502	-773	182	293	418	113	2,076**	-2,394**
	(1,370)	(767)	(104)	(303)	(572)	(280)	(728)	(882)
1	4,623**	-210	294*	368	813	213	2,520**	-3,073**
	(1,271)	(678)	(117)	(280)	(594)	(270)	(689)	(827)
2	5,411**	-148	314**	158	278	290	2,946**	-3,357**
	(1,320)	(720)	(102)	(280)	(519)	(285)	(741)	(834)
3	6,347**	-622	332**	110	945	822	3,243**	-2,641*
	(1,450)	(683)	(102)	(222)	(748)	(424)	(727)	(1,121)
4	5,933**	-842	352**	32	435	839*	3,728**	-3,481**
	(1,428)	(714)	(107)	(207)	(707)	(327)	(749)	(885)
5	7,115**	-1134	373**	195	1200	1,230**	4,324**	-3,730**
	(1,597)	(790)	(101)	(233)	(1,030)	(395)	(768)	(944)
6	8,669**	-67	376**	100	1480	1,546**	3,932**	-3,639**
	(1,670)	(791)	(110)	(247)	(1,157)	(401)	(728)	(1,076)
7	8,255**	-97	402**	123	1059	1,662**	4,449**	-3,442**
	(1,642)	(819)	(114)	(230)	(1,028)	(464)	(789)	(1,171)
8	8,451**	-281	391**	212	1553	1,937**	3,937**	-4,550**
	(1,644)	(826)	(116)	(225)	(1,068)	(473)	(777)	(1,020)
9	7,415**	-1492	436**	75	1038	2,073**	4,374**	-4,905**
	(1,710)	(839)	(116)	(247)	(1,056)	(470)	(821)	(1,078)
10	7,265**	-1,865*	467**	14	1097	2,734**	4,421**	-5,210**
	(1,707)	(797)	(103)	(240)	(1,073)	(490)	(812)	(993)

Appendix Table 8: Changes in Amount of Public Transfers (in 2016 Dollars) received Relative to Baseline, Chronic-Severe Disabled Women with Year of Onset prior to 1990

Year from Onset	Hours	Fraction not working	Earnings	Pre-tax Pre- transfer Income	After-tax Income without Transfers	After-tax income with non-SSA Transfers	After-tax income with SSA Transfers	After Tax Income with all Transfers	Food Consumption	Housing Consumption	Food plus Housing Consumption
-5	31	0.16	2.48	-1.48	-1.36	-0.72	-1.49	-0.4	1.03	-4.61*	-2.2
	(29)		(2.88)	(2.89)	(2.49)	(2)	(2.47)	(2.40)	(2.22)	(2.34)	(1.67)
-4	-33	0.16	-2.26	-0.46	-1.03	0.09	0.11	0.94	1.15	-4.24	-2.13
	(34)		(3.19)	(3.03)	(2.67)	(3)	(2.99)	(2.87)	(1.57)	(2.36)	(1.67)
-3	-32	0.16	-2.4	-3.05	-2.67	-2.37	-2.12	-1.89	2.04	-2.66	-0.56
	(31)		(2.91)	(2.98)	(2.66)	(3)	(2.90)	(2.81)	(2.62)	(3.69)	(2.62)
-2	-73*	0.18	-4.1	-2.12	-2.66	-3.99	-3.23	-3.02	-0.49	-1.78	-1.91
	(36)		(3.28)	(3.50)	(2.98)	(3)	(3.58)	(3.38)	(1.79)	(3.37)	(2.28)
-1	-191**	0.22	-11.33**	-5.58	-6.39*	-5.74*	-5.97*	-4.78	-2.9	-4.69	-4.06
	(34)		(3.01)	(3.47)	(2.70)	(3)	(2.93)	(2.81)	(2.10)	(3.80)	(2.61)
0	-239**	0.23	-15.13**	-0.13	-2.93	-2.26	-1.92	-0.83	-2.1	-5.85*	-4.49*
	(40)		(3.52)	(7.58)	(5.56)	(6)	(6.19)	(5.97)	(1.70)	(2.86)	(2.06)
1	-340**	0.32	-23.24**	-9.51**	-9.72**	-7.67**	-7.63**	-5.38	-0.62	-2.51	-1.91
	(38)		(3.13)	(3.66)	(2.89)	(3)	(2.95)	(2.84)	(2.06)	(3.67)	(2.48)
2	-317**	0.32	-24.56**	-9.99*	-10.66**	-7.83*	-8.39*	-5.29	-1.05	-7.72**	-5.67**
	(50)		(3.79)	(4.36)	(3.45)	(4)	(3.50)	(3.55)	(1.72)	(2.92)	(2.12)
3	-320**	0.33	-25.49**	-9.66**	-9.01**	-7.88**	-5.85*	-4.49	-1.3	-6.39	-4.82*
	(41)		(3.30)	(3.20)	(2.75)	(3)	(2.79)	(2.68)	(1.99)	(3.28)	(2.34)
4	-283**	0.27	-20.81**	-1.67	-3.72	-3.41	-0.81	-0.49	-1.22	-8.35**	-6.47**
	(55)		(3.92)	(6.24)	(4.78)	(5)	(4.85)	(4.68)	(1.88)	(2.69)	(2.01)
5	-336**	0.32	-24.11**	-9.68*	-9.58**	-8.63*	-6.25	-5.22	-1.64	-4.32	-4.14
	(43)		(3.60)	(4.53)	(3.64)	(4)	(3.68)	(3.55)	(2.27)	(3.75)	(2.64)
6	-302**	0.24	-24.85**	-0.55	-4.93	-4.47	-2.95	-2.54	-1.3	-8.29**	-6.32**
	(63)		(4.57)	(6.69)	(4.85)	(5)	(4.82)	(4.67)	(2.22)	(2.87)	(2.30)
7	-354**	0.30	-20.50**	-0.89	-2.87	-2.93	0.47	0.65	-2.07	-8.09*	-7.15**
	(47)		(4.13)	(6.37)	(4.75)	(5)	(4.58)	(4.41)	(2.73)	(3.44)	(2.59)
8	-378**	0.22	-27.21**	-10.04	-10.26*	-8.51	-7.87	-6.06	1.71	-10.45**	-6.93**
	(70)		(4.40)	(6.07)	(4.71)	(5)	(4.73)	(4.66)	(2.47)	(3.22)	(2.59)
9	-371**	0.31	-21.60**	-4.72	-5.19	-4.71	-2.14	-1.44	-1.21	-9.22*	-8.03**
	(54)		(5.40)	(4.74)	(3.95)	(4)	(3.88)	(3.71)	(3.68)	(4.04)	(2.96)
10	-396**	0.28	-29.30**	-13.49**	-13.05**	-12.61**	-10.63*	-10.14*	-0.17	-8.46*	-6.42*
	(76)		(4.90)	(4.93)	(4.13)	(4)	(4.14)	(4.02)	(2.77)	(3.61)	(2.89)

Appendix Table 9: Changes in Economic Outcomes Relative to Baseline, All-Disabled Women with Year of Onset after 1990

	All		SNAP/Food	Unemployment	Workers'	Supplemental	Social Security	Federal Income
Year from	Public	AFDC/TANF	Stamps	Insurance	Compensation	Security Income	(OASI and	Tax
Onset	Transfers		Stamps	mburunee	compensation	Security meenie	SSDI)	1 6/1
-5	1,041*	19	16	369	29	-12	334	-18
	(438)	(94)	(41)	(278)	(190)	(56)	(187)	(840)
-4	1,194**	54	15	-50	196	41	527**	406
	(461)	(146)	(34)	(110)	(285)	(65)	(194)	(1,090)
-3	652*	21	-20	-26	37	-94	578**	-710
	(325)	(105)	(36)	(91)	(169)	(57)	(198)	(955)
-2	976*	188	95*	-150	-139	-74	883**	138
	(412)	(164)	(39)	(114)	(211)	(94)	(238)	(1,213)
-1	1,650**	112	141**	-49	297	68	786**	-550
	(393)	(129)	(44)	(115)	(202)	(89)	(229)	(1,415)
0	1,998**	246	186**	55	103	185	934**	1668
	(458)	(201)	(45)	(129)	(237)	(104)	(229)	(2,546)
1	3,338**	125	195**	16	1,000**	318**	1,391**	-1226
	(464)	(126)	(49)	(125)	(317)	(104)	(255)	(1,407)
2	4,468**	158	199**	221	1488	224	1,721**	-1195
	(1,353)	(167)	(53)	(226)	(1,073)	(125)	(298)	(1,437)
3	3,336**	78	281**	-80	267	369**	2,061**	-1946
	(475)	(155)	(63)	(126)	(220)	(112)	(302)	(1,248)
4	2,690**	15	144**	-70	169	244*	2,033**	980
	(524)	(172)	(53)	(130)	(243)	(97)	(353)	(2,175)
5	3,246**	60	133*	-3	186	430**	2,103**	-1987
	(500)	(141)	(53)	(130)	(225)	(132)	(283)	(1,666)
6	2,164**	-91	72	-35	202	181	1,484**	2345
	(532)	(177)	(69)	(154)	(254)	(102)	(336)	(3,415)
7	3,140**	-152	282**	-168	36	429**	2,402**	664
	(525)	(142)	(101)	(127)	(207)	(149)	(363)	(2,788)
8	3,512**	-60	293**	215	802	236	1,826**	-2050
	(808)	(192)	(84)	(218)	(608)	(162)	(402)	(2,203)
9	3,103**	-112	212**	120	-10	443**	2,094**	-150
	(557)	(132)	(79)	(209)	(193)	(163)	(388)	(1,550)
10	2,226**	-190	236**	-108	100	296	1,684**	-4,198*
	(623)	(205)	(91)	(146)	(358)	(160)	(434)	(1,897)

Appendix Table 10: Changes in Amount of Public Transfers Received Relative to Baseline, All-Disabled Women with Year of Onset after 1990

Year		Fraction		Pre-tax Pre-	After-tax	After-tax income	After-tax	After Tax	Food	Housing	Food plus
from		not		transfer	Income without	with non-SSA	income with	Income with	Consumption	Consumption	Housing
Onset	Hours	working	Earnings	Income	Transfers	Transfers	SSA Transfers	all Transfers	Consumption	Consumption	Consumption
-5	-136	0.19	-9.81	-13.66*	-12.82*	-10	-10.37	-8.89	-3.95	-4.52	-5.21
	(99)		(8.10)	(6.91)	(6.45)	(6)	(6.70)	(6.15)	(8.70)	(5.55)	(4.91)
-4	-109	0.17	-14.04	-11.31*	-10.26*	-9	-7.95	-7.47	-6.49	-12.33*	-10.43**
	(111)		(7.70)	(5.15)	(4.82)	(5)	(5.47)	(4.93)	(4.34)	(5.12)	(3.85)
-3	-286**	0.19	-17.83*	-12.86*	-10.44*	-12.37**	-11.68*	-12.37**	21.55	-3.66	7.04
	(99)		(8.43)	(5.49)	(4.80)	(5)	(4.93)	(4.45)	(19.49)	(5.64)	(9.53)
-2	-161	0.23	-6.69	-1.46	-2.11	-8	-0.29	-6.87	2.42	-8.36	-5.05
	(116)		(10.12)	(6.13)	(5.85)	(6)	(7.21)	(5.99)	(5.97)	(6.38)	(4.84)
-1	-545**	0.31	-34.34**	-25.24**	-24.20**	-23.36**	-24.24**	-21.62**	-3.59	-12.38*	-9.74*
	(110)		(8.33)	(5.75)	(5.04)	(5)	(5.26)	(4.95)	(6.94)	(5.16)	(4.68)
0	-461**	0.30	-34.79**	-22.83**	-21.27**	-22.24**	-20.57**	-19.25**	-4.59	-15.62**	-12.08**
	(124)		(9.35)	(5.65)	(5.30)	(6)	(5.36)	(5.40)	(5.66)	(5.78)	(4.04)
1	-1,141**	0.73	-76.26**	-37.06**	-35.25**	-30.59**	-26.40**	-22.56**	-0.49	-12.62*	-9.23
	(105)		(6.45)	(5.04)	(4.74)	(5)	(4.78)	(4.59)	(7.10)	(5.84)	(4.89)
2	-1,059**	0.68	-70.93**	-30.75**	-29.99**	-27.18**	-19.71**	-18.23**	-4.85	-16.51**	-13.38**
	(182)		(8.98)	(6.98)	(6.68)	(7)	(6.54)	(6.96)	(5.79)	(6.06)	(4.30)
3	-1,209**	0.78	-84.33**	-40.70**	-38.10**	-32.72**	-25.39**	-20.97**	-5.61	-14.12**	-12.58**
	(104)		(4.48)	(6.17)	(5.99)	(6)	(5.80)	(5.23)	(6.33)	(4.76)	(3.67)
4	-1,212**	0.73	-80.85**	-27.18**	-26.80**	-22.82**	-10.03	-8.37	-8.48	-17.76**	-16.04**
	(145)		(7.17)	(9.68)	(9.13)	(8)	(8.43)	(7.63)	(6.12)	(6.38)	(4.48)
5	-1,300**	0.81	-88.95**	-44.40**	-41.83**	-37.93**	-27.59**	-24.68**	-0.29	-5.71	-6.65
	(97)		(3.56)	(5.21)	(5.05)	(5)	(4.70)	(4.65)	(5.13)	(9.06)	(4.71)
6	-1,526**	0.72	-93.80**	-37.81**	-35.06**	-29.50**	-21.09*	-17.65*	-5.05	-15.24*	-12.66*
	(154)		(3.09)	(8.50)	(8.58)	(8)	(8.35)	(8.12)	(5.38)	(7.06)	(5.01)
7	-1,325**	0.82	-86.96**	-42.95**	-40.41**	-38.36**	-24.28**	-23.23**	-13.76	-12.19	-17.00*
	(113)		(4.56)	(5.47)	(5.33)	(5)	(4.95)	(4.53)	(10.37)	(9.40)	(7.58)
8	-1,392**	0.62	-90.75**	-42.92**	-39.86**	-38.43**	-24.69*	-24.85*	-7.91	-15.15	-10.00*
	(243)		(4.88)	(11.22)	(10.31)	(10)	(11.62)	(10.85)	(5.16)	(8.10)	(4.87)
9	-1,311**	0.82	-87.44**	-40.26**	-39.67**	-39.44**	-25.07**	-25.91**	-4.33	-22.60**	-20.48*
	(121)		(4.20)	(8.95)	(7.89)	(7)	(6.74)	(6.21)	(14.59)	(7.83)	(8.49)
10	-1,862**	0.90	-98.47**	-54.57**	-50.27**	-48.58**	-35.22**	-35.12**	-5.89	-11.58	-7.63
	(107)		(1.33)	(11.91)	(11.50)	(11)	(13.54)	(12.07)	(7.96)	(8.08)	(6.30)

Appendix Table 11: Changes in Economic Outcomes Relative to Baseline, Chronic-Severe Disabled Women Year of Onset after 1990

Year from Onset	All Public Transfers	AFDC/TANF	SNAP/Food Stamps	Unemployment Insurance	Workers' Compensation	Supplemental Security Income	Social Security (OASI and SSDI)	Federal Income Tax
-5	936	97	102	-13	887	29	94	-1746
	(1,250)	(260)	(167)	(272)	(856)	(213)	(533)	(983)
-4	1211	-153	-80	-500*	1295	-61	720*	-1441
	(1,131)	(373)	(103)	(225)	(1,019)	(102)	(353)	(1,019)
-3	-695	-253	-307**	-247	207	-316	336	-3,616**
	(750)	(245)	(104)	(315)	(366)	(263)	(461)	(1,305)
-2	-1,887	-862	-44	-542**	-1156	-406	1257	-208
	(1,425)	(472)	(116)	(164)	(911)	(286)	(858)	(855)
-1	2020	38	63	-203	1184	155	826	-3,640*
	(1,337)	(299)	(144)	(294)	(916)	(476)	(601)	(1,467)
0	2351	841	45	166	-689	-114	1,838*	-3,506**
	(2,007)	(982)	(113)	(593)	(863)	(428)	(832)	(1,078)
1	6,905**	196	396*	-312	1403	453	4,437**	-4,824**
	(1,526)	(311)	(184)	(299)	(1,091)	(494)	(1,024)	(1,212)
2	6,183**	-82	94	-551*	836	577	5,335**	-3,402**
	(2,075)	(503)	(167)	(216)	(1,553)	(351)	(1,172)	(1,051)
3	9,569**	175	92	274	1401	684	6,808**	-5,862**
	(1,773)	(336)	(171)	(517)	(1,214)	(368)	(1,393)	(1,389)
4	10,153**	537	199	-519*	1139	1,409*	7,345**	-2277
	(2,480)	(322)	(143)	(233)	(1,567)	(696)	(2,094)	(1,280)
5	9,621**	937	140	-510*	687	1,477**	7,108**	-6,589**
	(1,781)	(518)	(162)	(201)	(1,076)	(490)	(1,159)	(1,387)
6	9,967**	1190	895	-457	320	650	6,834**	-5,908**
	(2,538)	(610)	(478)	(358)	(600)	(584)	(2,145)	(1,819)
7	10,445**	-148	633*	-871**	-82	2,152**	8,191**	-5,990**
	(1,906)	(344)	(271)	(193)	(942)	(795)	(1,469)	(1,751)
8	7,671**	422	627**	-887**	-655*	-4	8,337**	-7,684**
	(1,900)	(343)	(229)	(220)	(334)	(119)	(1,750)	(2,932)
9	7,702**	156	655*	-386	-1287	2,024*	6,743**	-3321
	(1,810)	(401)	(257)	(269)	(681)	(913)	(1,378)	(1,900)
10	7,815**	391	691	-1,062**	-261	552	7,624*	-13,294**
	(2,710)	(459)	(630)	(252)	(319)	(514)	(2,984)	(3,380)

Appendix Table 12: Changes in Amount of Public Transfers (in 2016 Dollars) received Relative to Baseline, Chronic-Severe Disabled Women with Year of Onset after 1990